IMPROVING THE MECHANICAL PERFORMANCE OF WOOD FIBER REINFORCED BIO-BASED POLYURETHANE FOAM

by

Li-Chi Chang

A thesis submitted in conformity with the requirement for the degree of Master of Applied Science Graduate Department of Chemical Engineering and Applied Science University of Toronto

© Copyright by Li-Chi Chang (2014)

Improving the Mechanical Performance of Wood Fiber Reinforced Bio-based Polyurethane Foam

Li-Chi Chang

Master of Applied Science Department of Chemical Engineering and Applied Chemistry

University of Toronto

2014

Abstract

Because of the environmental impact of fossil fuel consumption, soybean-based polyurethane (PU) foam has been developed as an alternative to be used as the core in structural insulated panels (SIPs). Wood fibers can be added to enhance the resistance of foam against bending and buckling in compression. The goal of this work is to study the effect of three modifications: fiber surface treatment, catalyst choice, and mixing method on the compression performance of wood fiber-reinforced PU foam. Foams were made with a free-rising process. The compression performance of the foams was measured and the foams were characterized using Fourier transform infrared spectroscopy (FTIR), scanning electron microscopy (SEM), and X-ray computed tomography (CT). The foam reinforced with alkalitreated fibers had improved compression performance. The foams made with various catalysts shared similar performance. The foam made using a mechanical stirrer contained well-dispersed fibers but the reinforcing capability of the fibers was reduced.

Acknowledgements

It is my pleasure to have the opportunity to work with Prof. Mohini Sain and Prof. Mark Kortschot. I really appreciate their supervision and guidance in the course of pursuing the degree. Special thanks go to members in Prof. Kortschot's research group: Billy Cheng, Numaira Obaid, and Sadakat Hussain, and those in Prof. Sain's research group: Mustafa Khazabi, Omar Fauk, Javad Sameni, Pei-Yu Kuo, and Ruijun Gu for their help and assistance in the development of this study. In addition, I would like to specially acknowledge the generous assistance from Tony Ung and Shiang Law. Financial support for this work was generously provided by FPInnovations (Robert M. Knudson) and NSERC. Last but not least, I would like to thank my brother and parents for all their support.

Table of Contents

Abstract	ii
Acknowledgements	iii
Table of Contents	iv
List of Tables	vii
List of Figures	viii
List of Nomenclature	xi
1. Introduction	1
2. Literature Review	3
2.1 Introduction	3
2.2 Polyurethane Foams	8
2.2.1 Introduction	8
2.2.2 Natural Resources for Foam Development	11
2.2.3 Soybean oil-based Polyol	12
2.3 Wood Fibers as Reinforcement in Composite Foams	16
2.3.1 Composite Foams	16
2.3.2 Polyurethane Composite Foams	19
2.3.3 Fiber/Polymer Interface	20
2.3.4 Role of Catalyst	
2.3.5 Fiber Dispersion	24
2.4 Problem Statement	25

2.5 Conclusions	6
3. Experimental Methods	7
3.1 Fiber-reinforced Foam Preparation27	7
3.1.1 Materials	7
3.1.2 Fiber Treatment	8
3.1.3 Mixing	0
3.1.4 Foaming Process	0
3.2 Fiber-reinforced Foam Characterization	4
3.2.1 Mechanical Testing	4
3.2.2 FQA and FTIR Characterization	4
3.2.3 Imaging Characterization	5
4. Results and Discussion	6
4.1 Introduction	6
4.2 Alkali Treated Fiber-reinforced PU Foams	7
4.2.1 Characterization of Treated Fiber	7
4.2.2 Compression Performance of Treated Fibre-reinforced Foam44	4
4.2.3 Cell Morphology	3
4.2.4 Conclusions	6
4.3 Effect of Catalyst on the Compression Performance of Foams	7
4.3.1 Reaction Profile of PU Foams	7
4.3.2 Effect of Catalyst on the Compression Behavior of Foam60	0
4.3.3 FTIR Characterization of PU Foams	5

4.3.5 Conclusions	73
4.4 Characterization of Foams Made Using Different Mixing Methods	74
4.4.1 Effect of Stirring Time on the Compression Response of Foam	74
4.4.2 Effect of Stirring Rate on the Compression Response of Foam	79
4.4.3 Mixing Method Comparison on the Compression Response of Fo	am 84
4.4.4 Conclusions	90
5. Conclusions and Recommendations	91
5.1 Conclusions	91
5.2 Recommendations	92
6. References	94

List of Tables

Table 2-1 - U.S. foam demand (total: 1.3 billion dollar) in 2007	8
Table 2-2 - General applications of PU foams	.11
Table 2-3 - PU foams with natural fiber made using different mixing methods	. 25
Table 3-1 - Materials used in the preparation of foam	.31
Table 3-2 - Catalysts used in foam production	. 32
Table 4-1 - Cream time, gel time and rise time of PU foam prepared with catalyst: CAT-0, CAT-1, CAT-2, CAT-3, and CAT-4.	. 59
Table 4-2 - Relative concentration of urea and urethane linkages of the foams made using catalysts, CAT-0, CAT-1, CAT-2, CAT-3, and CAT-4	.70

List of Figures

Figure 2-1 -	U.S. market of polymeric foams7
Figure 2-2 -	• PU synthesis reaction (elements parenthesized mean double bond of element at branch point in a polymer)
Figure 2-3 -	Thermal conductivity (λ) of cellular materials
Figure 2-4 -	Chemical structure of soybean oil-based polyol
Figure 2-5 -	Energy sources for polyol productions
Figure 2-6 -	Acid gas emissions for polyol productions15
Figure 2-7 -	Greenhouse gas emissions for polyol productions
Figure 2-8 -	Blowing reaction of the synthesis of PU foam
Figure 3-1 -	Photos of (a) unstained fibers and (b) stained fibers
Figure 3-2 -	Stereo micrographic images of (a) unstained fibers and (b) stained fibers 29
Figure 3-3 -	Schematic for the preparation of fiber-reinforced foam
Figure 3-4 -	Schematic for the preparation of fiber-reinforced foam using a mixing stirrer33
Figure 4-1 -	Distribution of length of (a) untreated and (b) treated fibers measured by FQA 38
Figure 4-2 -	Distribution of diameters of (a) untreated and (b) treated fibers
Figure 4-3 -	SEM images of fiber surface of (a) untreated and (b) treated fibers
Figure 4-4 -	SEM micrographs of (a) untreated fibers and fibers treated in solutions of (b) 2, (c) 4, and (d) 7 wt% NaOH
Figure 4-5 -	FTIR spectra of (a) untreated and (b) treated fibers
Figure 4-6 -	Compressive strengths of foams with the reinforcement of untreated fibers (unWF) and 2.5 wt % of fibers treated with 2 wt% (02WF), 4 wt% (04WF), and 7 wt% (07WF) NaOH solutions
Figure 4-7 -	Specific strengths of foams with the reinforcement of untreated fibers (unWF) and 2.5 wt % of fibers treated with 2 wt% (02WF), 4 wt% (04WF), and 7 wt% (07WF) NaOH solutions
Figure 4-8 -	• Compressive strengths of untreated fiber-reinforced (white) and treated fiber- reinforced (grey) foams with various fiber contents from 0 to 10 wt% with an interval of 2.5 wt%

Figure 4-9 - Specific compressive strengths of untreated fiber-reinforced (white) and treated	
fiber-reinforced (grey) foams with various fiber contents from 0 to 10 wt% with an interval of 2.5 wt%	8
Figure 4-10 - Experimental modulus of untreated ($E_{c(untreated)}$) and treated ($E_{c(treated)}$) fiber- reinforced foams and modulus of the matrix for composite foam of untreated ($E_{x(untreated)}$) and treated ($E_{x(treated)}$) fiber-reinforced foams with different fiber contents	0
Figure 4-11 - Specific modulus of untreated (E _{sp(untreated)}) and treated (E _{sp(treated)}) fiber- reinforced foams with different fiber contents	2
Figure 4-12 - Reinforcing factors of untreated fiber-reinforced and treated fiber-reinforced foams as a function of fiber content	3
Figure 4-13 - Reconstructed X-ray tomographic image of neat foam	5
Figure 4-14 - Reconstructed X-ray tomographic images of foams with (a) 2.5, (b) 5, (c) 7.5, and (d) 10 wt%	б
Figure 4-15 - Chemical structures of (a) ethanolamine (CAT-0), (b) benzyldimethyl amine (CAT-1), (c) 2-(2-aminoethoxy) ethanol (CAT-2), (d) 2-(methylamino) ethanol (CAT-3), and (e) 2-[2-(dimethylamino)ethoxy] ethanol (CAT-4)59	8
Figure 4-16 - Rise height of PU foams prepared with catalyst: (a) CAT-0, (b) CAT-1, (c) CAT-2, (d) CAT-3, and (e) CAT-4 as a function of time	0
Figure 4-17 - Measured compressive strengths of PU foams made with CAT-0, CAT-1, CAT-2, CAT-3, and CAT-4	1
Figure 4-18 - Measured compressive moduli of PU foams made with CAT-0, CAT-1, CAT-2, CAT-3, and CAT-4	2, 2
Figure 4-19 - Densities of PU foams produced with various catalysts (CAT-0, CAT-1, CAT- 2, CAT-3, and CAT-4)	3
Figure 4-20 - Specific compressive moduli of PU foams produced with catalysts: CAT-0, CAT-1, CAT-2, CAT-3, and CAT-4	4
Figure 4-21 - Schematic diagram of the activation mechanism of isocyanate-hydroxyl gelling reaction	; 5
Figure 4-22 - FTIR spectra of PU foams prepared with CAT-0, CAT-1, CAT-2, CAT-3, and CAT-4	7

Figure 4-23 - Cu 135	rve-fitted FTIR spectrum of the foam made using CAT-0 as a catalyst in the 50-1800 cm ⁻¹ region	9
Figure 4-24 - SE (d)	M micrographs of PU foams made with (a) CAT-0, (b) CAT-1, (c) CAT-2, CAT-3, and (e) CAT-4 in the foam rising direction	2
Figure 4-25 - Co	ompressive strength of fiber-reinforced foam as a function of stirring time7	5
Figure 4-26 - Co	ompressive modulus of fiber-reinforced foam as a function of stirring time. 70	б
Figure 4-27 - Re	inforcing factor of fiber-reinforced foam as a function of stirring time7	7
Figure 4-28 - SE	M micrographs of foams made with (a) 10 and (b) 30 min stirring times78 \times	8
Figure 4-29 - Re min	constructed X-ray tomographic images of foams made with (a) 10 and (b) 30 n stirring times) 9
Figure 4-30 - Co	mpressive strength of fiber-reinforced foam as a function of stirring time8	0
Figure 4-31 - Co	mpressive modulus of fiber-reinforced foam as a function of stirring rate8	1
Figure 4-32 - Re	inforcing factor of fiber-reinforced foam as a function of stirring rate8	1
Figure 4-33 - SE	M micrographs of foam made with 600 rpm stirring rate83	3
Figure 4-34 - Re rate	constructed X-ray tomographic image of foam made with 600 rpm stirring	3
Figure 4-35 - Co (gr	ompressive strengths of foams made using hand (white) and mechanical ey) mixings	5
Figure 4-36 - Co miz	mpressive moduli of foams made using hand (white) and mechanical (grey) xings	5
Figure 4-37 - Re fur	inforcing factors of foams made using hand and mechanical mixing as actions of fiber content	б
Figure 4-38 - SE and	M micrographs of foams containing 10 wt% of fibers made using (a) hand 1 (b) mechanical mixing	8
Figure 4-39 - Re ma	constructed X-ray tomographic images of foam containing 10 wt% of fibers de using (a) hand and (b) mechanical mixing	8
Figure 4-40 - Ste har	ereo microscopic images of foams containing 5 wt% of fibers made using (a) and (b) mechanical mixing	9

List of Nomenclature

C_1		Geometric constant of proportionality
D	μm	Diameter of fiber
Ec	MPa	Experimental modulus of foam sample
Eo	MPa	Elastic modulus of the foam with no fibers
E_{PU}	MPa	Elastic modulus of cell wall material (PU)
E _x	MPa	Modulus of the matrix for composite foam with the removal of
		fibers
L	mm	Length of fiber
m _{fibers}	g	Mass of fibers
m _{foam}	g	Mass of the entire foam sample
m _{solid}	g	Mass of cell struts
V _{foam}	cm ³	Volume of the entire foam sample
$\mathbf{V}_{\text{solid}}$	cm ³	Volume of cell struts
$ ho_o$	g/cm ³	Density of the foam without fibers
$ ho_{PU}$	g/cm ³	Density of cell wall material (PU)
ρ_x	g/cm ³	Density of the matrix for composite foam with the removal of
		fibers

1. Introduction

Polyurethane foam is a synthesized cellular plastic and has been used in a number of diverse fields, such as cushioning, packaging, automotive, and structural applications. In the field of structural applications, polyurethane foam is an insulating core material placed in structural insulated panels (SIPs); SIPs consist of two sheathing layers with an insulating core. Polyurethane foam being used as a core material needs to have the ability to withstand compressive and shear loading.

In order to enhance the mechanical properties of foam without an increase in petrochemical feedstock, polyurethane foams reinforced with wood fibers have been developed. The thermal stability, energy absorption, physical and mechanical properties of wood fiber-reinforced polyurethane foam have been studied in the literature. Although considerable effort has been spent in studying wood fiber-reinforced polyurethane foams, one problem is still unsolved: compressive strength and modulus of foam either decreases or stays constant with increasing fiber content. This study was carried out to understand and enhance the compression performance of wood fiber-reinforced polyurethane foam. The results showed that the application of alkaline treatment and the use of a mechanical stirrer enhanced the compression performance of fiber-reinforced foam.

The main objective of this study is to try several methods to enhance the compression performance of wood fiber-reinforced polyurethane foam. In order to achieve the objective, approaches centering on the surface modification of fiber and manufacturing process of foam were implemented. In the thesis, three aspects of foam performance were investigated: (1) the effect of fiber surface treatment, (2) the effect of catalyst, and (3) the effect of mixing method.

There are five chapters in the thesis. Chapter 1 gives an overview of the thesis and Chapter 2 provides an introduction to polyurethane foam and other plastic foams and covers related works reported in the literature. Chapter 3 describes the experimental methods, and materials used during the experiments. Chapter 4 presents experimental results along with the discussion of the results. Lastly, in Chapter 5, the results are summarized and recommendations proposed for future work.

2. Literature Review

2.1 Introduction

Foam is a three-dimensional cellular solid containing pore space and a skeletal frame. Foam is found widely in nature and is used extensively by humans; it is primarily found in structural applications. The mechanical behaviors of foam depend on cellular structure. The reduction of weight and increase in load-carrying capability under bending are advantages of the cellular structure. Wood and cancellous bone are two examples of natural cellular materials taking advantage of this.

Wood is composed of lattices of cells, basic units of wood structure. These cells act as vessels for fluid conduction and provide structural support; cell vessels facilitate the transportation of water and the foam-like structure of wood allows it to resist bending in order to maximize the growing efficiency of tree [1].

Skull bone is a sandwich structure composed of layers of compact bone with internal cancellous bone as a core. Cancellous bone is a porous solid composed of interconnected cells. The cellular structure of the cancellous bone is responsible for separating layers of dense bone and increasing the moment of inertia [13].

Cellular materials with low weight and high rigidity have been exploited in many applications and served human for millennia. Wood, for example, has been used extensively as construction materials due to its light weight and high stiffness. For its application in structural engineering, low density wood has been used in sandwich panels as core material which was replaced by other materials such as honeycombs and plastic foams after World War II and high density wood is used as external layer material. The need of alternative lightweight and rigid foamed materials has grown and leads to the development of synthetic foams.

Foams also can be made artificially. Many materials can be foamed, for instance, glasses, ceramics and metals. In particular, plastic foam is of academic and industrial interest. Plastic foam consists of a polymer matrix filled with numerous pores. Plastic foams such as polyurethane (PU) foam, polyolefin foam and polystyrene (PS) foam are commonly found in daily use and industry.

The first man-made cellular polymer, sponge rubber, was made in 1914 [3]. The first synthetic plastic foam, extruded PS foam, was developed in 1931[4, 5]. In 1937, Otto Bayer synthesized the first PU foam [6]. After the war, the increase in demand of foam from industry caused the widespread use of rigid PU foams [7-10]. Flexible PU foam was introduced into market in the early 1960s [11].

Foaming is a process of generating and stabilizing gas bubbles in polymeric solid. Foaming mechanisms can be divided into two categories: soluble foaming and reaction foaming. In soluble foaming, blowing agents are mixed with polymer and release of gas is controlled by either the decrease in pressure or increase in temperature. Examples of this category include polyethylene (PE) foam, polypropylene (PP) foam, polyvinyl chloride (PVC) foam and PS foam made by extrusion and injection molding. In reaction foaming, the release of gas is governed by the reaction of the blowing agent with other chemicals or thermal decomposition of blowing agent. Generally, PU foam, polyisocyanurate (PIR) foam, and phenolic foam are fabricated through reaction foaming [12]. Different formulations and processing variables in foam fabrication result in different cell shape distributions and cellular structures. The change in structure affects physical, thermal, and mechanical properties of the foam.

Plastic foams are classified according to thermo-reversibility of a polymer and the cellular structure of foam. Plastic foams can be classified into thermoplastic foam and thermosetting foam according to the thermo-reversibility or be divided into closed-cell foam and open-cell foam according to the cell structure of foam [12].

For open-cell foam, the cell walls are ruptured and air fills the hollow cells. In contrast, cell walls of closed-cell foam are essentially intact and cells are filled with gas. The gas is either remaining gas from those used to saturate polymer or gas released by blowing agent during the formation of foam [13]. Open-cell foam has less inherent resistance under external force due to ruptured cell wall and a lower R-value because of the interconnected pores. On the other hand, the high rigidity and R-value of closed-cell foam are attributed to the support of cell walls and insulation of gas enclosed in foam cells, respectively; however, disadvantages of closed-cell foam are high cost and density compared with those of open-cell foam. The choice between open-cell and closed-cell foams depends on the application. Generally, open-cell foams have been used to develop products in shock absorption and cushioning package while closed-cell foams are applied for insulation purposes in building construction.

Thermoplastic foams including PE foam, PP foam, PS foam, and PVC foam are used in many applications. PE foams are used in agriculture, automotive, packaging and building industry. PE foams are also found in the insulation of heating pipe and enhancing the resilience of sport protector [14, 15]. PP foam has high abrasion resistance and substitutes

- 5 -

other thermoplastic foams in a lot of application domains [16]; PP foams are commonly used in food manufacturing and automotive industry. It is because PP foams have higher stiffness compared with PE foam and larger impact strength with respect to PS foam. Most importantly, production cost of PE and PP foams is lower than that of PS foam. In addition, expanded PP foam developed by JSP Corporation [17] is considered as a suitable material for flotation devices, such as life jacket and buoyancy aid [18]. PVC foam is another cellular material commonly used in flotation applications. It is used in the marine and shipbuilding industry for its high tolerance to dynamic loading in the marine environment [19, 20]. PS foam is used in packaging and building construction [21]. In construction, open-cell PS foam is used as a major core material in vacuum insulated panel (VIP) for thermal insulation [22-24].

Thermosetting foams are also suitable materials for construction application. For instance, PIR foam has been used not only in domestic hot water storage tanks but also in construction [25]. It was reported that PIR foam has better fire resistance than other plastic foams, making it a desirable material to be applied in construction; this is because urethane linkages in PU foam are replaced by imide or amide linkages [26] resulting in improved thermal stability but with high surface friability [27, 28]. In addition to the high friability causing failure at the foam/sheathing interface, PIR foams have another drawback, a high content of isocyanate used for the synthesis of foam. Without these disadvantages and with excellent thermal insulating characteristics, PU foams have been used for insulation in building; they are well known core materials for structural insulated panels (SIPs) in construction.

The market for plastic foam is expanding rapidly. According to a BCC research report [29] focusing on the U.S. market, 7.5 billion pounds of plastic foams were produced 2010. The report predicted that this amount will increase to 8.6 billion pounds with an annual growth rate of 2.9 %. 443 million pounds of polyolefin foams, including PE foam and PP foam, was produced in 2010 and the amount of polyolefin foams is expected to grow by 15 % from its 2010 level to 509 million pounds by 2015.

The largest market share in plastic foam industry belongs to PU foam. 3.9 billion pounds of PU foam was produced in 2010 and the amount of PU foam is forecast to be 4.4 billion pounds in 2015 with an annual growth rate of 2.6%. Figure 2-1 shows the past and estimated market for plastic foams. Another report done by Freedonia group (Cleveland) described U.S. foam demand in 2007 (Table 2-1). The demand analysis predicted an expanding market for PU foam [30].



SUMMARY FIGURE POLYMERIC FOAM U.S. MARKET ESTIMATES BY RESIN FAMILY, 2009-2015 (MILLION LBS.)

Source: BCC Research

Figure 2-1 U.S. market of polymeric foams [29]

Materials	Percentage (%)
Polyurethane Foams	33
Engineering Plastic Foams	20
Non-plastic Foams	17
Polyolefin Foams	10
Others (Metal and Ceramic foams)	20

Table 2-1 U.S. foam demand (total: 1.3 billion dollar) in 2007 [30]

2.2 Polyurethane Foams

2.2.1 Introduction

PU is a synthetic polymer where organic units are formed by connecting molecules with urethane linkages [31]. The synthesis of PU is mainly based on the reaction between isocyanate group of di-isocyanate and hydroxyl group of polyol (Figure 2-2 shows this reaction). End groups of the polymer chain are determined by dominant species among reactants; an excess of di-isocyanate results in the polymer chain being end capped with isocyanate groups.

$$O=C=N-R_{(isocyanate)}-N=C=O + HO-R_{(polyol)}-OH \rightarrow O=C=N-R_{(isocyanate)}-NH-C(O)-O-R_{(polyol)}-O-C(O)-NH-R_{(isocyanate)}-N=C=O$$

Figure 2-2 PU synthesis reaction (elements parenthesized mean double bond of element at branch point in a polymer)

Other reactions may accompany the formation of PU. For instance, isocyanate groups, in addition to reacting with the hydroxyl groups of the polyol, can react with the urethane linkages of PU to form allophanate. In addition, the isocyanate group can react with other isocyanate groups to form dimers or trimers. With the presence of water, the isocyanate group can react with urea to form biuret, where urea is a product of the reaction between water and isocyanate. Although these side reactions lower the stability of PU, at the same time they can increase the degree of crosslinking between molecules, increasing the stiffness of the PU [32].

Based on the degree of crosslinking, PU foams are divided into three groups: flexible foam, semi-rigid foam, and rigid foam. They are made from different formulations and with different compositions. In general, flexible foams are used for packaging and cushioning, while rigid foams are for building insulation. Table 2-2 lists common application of different types of PU foams.

PU has been developed for versatile applications, such as paints, coatings, adhesives, elastomers, and foams. The high degree of vapor permeability and water resistance of PU film enables it to be used for flexible surface coating. PU adhesives have been extensively

used for bonding between core material and sheathing in structural panels as well as between interiors in automotive vehicles [33, 34]. For implanting medical devices, PU elastomer outperforms silicon elastomer with regards to its high abrasion and fatigue resistance under dynamic loading [35]. Insulation is one of them. PU foams have advantages of lightweight, low cost, and especially good adherence toward the surface of sheathing layer. More importantly, they are popular insulating materials with excellent insulating properties for their lowest thermal conductivity over other cellular materials (shown in Figure 2-3). Hence, PU foams take up a considerable fraction of PU production (almost a quarter) [36].



Figure 2-3 Thermal conductivity (λ) of cellular materials [34]

Foams	Markets	Applications	
Flexible PU	Transportation	Seating, pads, liners, dampening, carpet backing, filters,	
		flooring, armrests, trim	
	Furniture	Bedding, padding, flooring	
	Recreation	Sport mats, toys, helmet liner, chest protection	
	Packaging	Electronic, computer, china, equipments	
Semi-rigid PU	Automotive	Dash panel, liner, viser	
	Footwear	Soles	
Rigid PU	Construction	Insulation, flooring, siding	
	Appliance	Refrigerator frame, door, dishwasher door	
Table 2-2 General applications of PU foams [37]			

2.2.2 Natural Resources for Foam Development

Polyol, which is a main reactant for commercially PU production, is based on polyethers obtained from the reaction between initiators and propylene oxide or ethylene oxide, both of which are derived from petrochemical sources [38].

Because of environmental concerns and unstable petroleum price profiles, there is a trend toward substituting petroleum-based polyol with renewable resources. The use of natural oils in the synthesis of polyol for PU foam preparation has been reported in the literature. For example, canola oil [39], castor oil [40], lesquerella oil [41], rape seed oil [42],

palm oil [43], and soybean oil [44] have all been used for this purpose. Furthermore, the fabrication of PU foam using bio-wastes recycled from pulp and paper mill was studied [45, 46], in an attempt to solve the problem of increasing paper wastes and reduce the energy consumption of PU foam production. Another way of producing environmentally friendly PU foam is to leverage the abundance of agricultural resources. Liquefaction of agricultural materials containing cornstalk [47], bamboo [48], sugar cane [49], bark [50], wood and starch [51] were conducted and reported for the preparation of PU foams in the literature. In addition, PU foams can also be made from lignin through oxypropylation [52, 53].

2.2.3 Soybean oil-based Polyol

Soybean oil-based polyol is derived from soybean oil through a continuous two-step process. Unlike caster and lesquerella oils, soybean oil lacks a hydroxyl group on the chain. Thus, the first step is to add the hydroxyl group to the side chain of soybean oil; in the synthesis of polyol, performic acid is added to form epoxidized soybean oil. Then epoxidized oil then undergoes ring opening and hydroxylation to form polyol. Figure 2-4 shows the chemical structure of soybean oil-based polyol [54].



Figure 2-4 Chemical structure of soybean oil-based polyol

Tan *et al.* compared the properties of foam made from soybean oil-derived polyol with those of foam made using petrochemical-based polyol. The results showed that the foam made from soybean oil-based polyol can reach comparable thermal conductivity and even have better compressive performance than that of the petrochemical counterpart. However, the thermal conductivity of soybean oil-based foam increases faster than that of petroleum-based foam under humid conditions [36].

Life cycle assessments have been implemented in order to estimate whether soybean oil-based polyol is a potential substitute for petrochemical-based polyol. One study [55] assessed the production systems of soybean oil and petrochemical polyols regarding their input, output and environmental impact. Figure 2-5 shows the composition of energy sources consumed by each system. For the productions of castor oil and soybean oil-based polyols, 35 % of energy is attributed to fossil fuel-related source while for the production of petrochemical-based polyol over 90 % of energy consumed comes from fossil fuel. When acid gas emissions of the production of polyol were taken into account, the production of

natural oil-based polyol emitted only one fifth of the amount that the production of petrochemical-based polyol did (Figure 2-6). Another important and beneficial aspect of using polyols derived from natural oils is on environmental impact; in Figure 2-7, both the polyols made from castor and soybean oils have essentially zero carbon dioxide emissions. These data confirmed that natural oil provides a valuable and renewable route with minor environmental impact for the synthesis of polyol.



Figure 2-5 Energy sources for polyol productions [55]



Figure 2-6 Acid gas emissions for polyol productions [55]



Figure 2-7 Greenhouse gas emissions for polyol productions [55]

2.3 Wood Fibers as Reinforcement in Composite Foams

Advanced composites are high-performance materials that benefit from high stiffness of fibrous materials embedded in polymeric matrix; fiber-reinforced resin composites and composite foam are examples. Fiber-reinforced composites are used in a wide range of applications such as aerospace, automobile, electronic, and marine industry [56]. Composite foams are mainly used as sandwich core materials for structural applications. The mechanical behavior of fiber-reinforced composites and composite foams depends not only on the properties of polymer matrix but also on those of fiber. In addition to increased stiffness, another advantage of adding fiber is to reduce the amount of material used in manufacturing and the weight of produced composites. Reduction of material usage can cause a decrease in feedstock cost and reduction in weight can lower carrying costs of composites in transit.

As environmental concerns become crucial in industry and 'green' products attract environmentally-conscious consumers, composites with natural fibers are displacing those with synthetic fibers. The application area and market of natural fiber-reinforced composites is expected to grow at a steady pace for future decade.

2.3.1 Composite Foams

Fiber-reinforced polymeric composites, a combination of fiber and polymer resin, has been commercialized and studied since 1916. In order to reduce processing and material costs, various kinds of fillers have been designed and added in thermoplastics. They are classified into two kinds: (1) inorganic fillers: glass fiber [57], calcium carbonate [58], talc and wollastonite [59], and (2) organic fillers: corn starch [60], rice hull [61], bamboo fiber [62], sisal fiber [63], kenaf fiber [64], wood flour and fiber [65].

Wood fibers in polymeric composites serve as fillers and reinforcing agents to achieve the desired characteristics of composites and thus are widely used in industry [66]. Wood fibers have high stiffness, low cost and natural fiber sources. However, wood fiber-reinforced composites have some limitations: reduced ductility, low resistance under impact and in moist conditions, increased brittleness, and high density compared with those of pure thermoplastics. In order to solve the above mentioned problems, composite foams were developed [67]. It was demonstrated that composite foams have enhanced fatigue life [120], impact strength, toughness, and most importantly, reduced density [121].

Wood fibers are added to composite foams as reinforcing fillers. The advantages of using the fibers are low cost, renewable sources, and small environmental burden. In addition, the density of the fibers is less than glass fibers and the fibers have better adhesive property between fibers and polymer matrix.

Yoon *et al.* made PE composite foams using injection molding with a physical blowing agent (PBA), nitrogen and compared PE solid composite and PE composite foam on the density, shrinkage, and mechanical properties. The results showed composite foams had reduced density and shrinkage, and enhanced mechanical properties [68]. Zhang *et al.* fabricated PE foam with wood fiber reinforcement using tandem extrusion method with carbon dioxide as a PBA. Comparing with the results of foam made by single extruder system, the authors described that fine-celled structure was observed of the foam made using tandem extrusion method; the fine structure of foam cells was attributed to the uniform mixing achieved by using tandem extrusion [69].

Bledzki *et al.* prepared wood fiber-reinforced PP foams by injection molding, compression molding, and extrusion with a chemical blowing agent (CBA) [70]. The effects of the CBA and flame retardant on mechanical properties of wood fiber-reinforced PP foam were studied by Bledzki *et al.* [71] and Zhang *et al.* [72], respectively. Furthermore, composite foams made by blending PE and PP with wood fibers were produced. It was observed that the foam made by blending PE and PP had uniform cell structure, while the individual cell morphology of PE foam and that of PP foam were unfavorable [73].

PVC composites with wood fibers for reinforcement have also been foamed and the dependence of cell morphology on the processing parameters was investigated [74, 75]. The relationship between mechanical properties and cell morphology was also studied. It was found that the impact strength of PVC/wood fiber foam increased with the increase of void fraction [76].

Wood fiber-reinforced PS foams were produced through extrusion and moisture in wood fibers was used as a blowing agent. One advantage of using wood fiber as a carrier of blowing agent is that moisture can be gradually released from fiber generating well-dispersed bubbles in polymer solid during the foaming process [77]. The concept of using moisture as a blowing agent has been extended to polypropylene carbonate (PPC) foam system [78].

Since the wastes of the foams described above (PE, PP, PVC and PS foams) are not biodegradable and become sources of pollution from disposal, foams based on biodegradable polylactic acid (PLA) have been investigated. Wood fiber-reinforced PLA composite foams have been developed using a batch process with supercritical carbon dioxide and improved stiffness of foams was reported [79]. It was shown that foam density, fiber-fiber interaction, and fiber-matrix interface are factors that can affect the mechanical behavior of fiberreinforced composite foams.

In order to analyze the behavior of wood fiber-reinforced foams, not only the properties of polymeric matrix but also the dimensions of fiber needs to be taken into account. This is because the size of fiber affects the surface area that is exposed to the matrix, and also affects the viscosity of the fiber-polymer blend and cell expansion mechanism. For example, in the manufacturing of PE foam containing different lengths of wood fibers but the same content of the fibers, a larger viscosity was observed in the foam reinforced with shorter fibers due to larger surface areas compared with longer ones. As the surface area increased, more sites were available for cell nucleation. Therefore, small cell size and fine cell morphology were obtained [80].

2.3.2 Polyurethane Composite Foams

There have been many studies of PU foams containing wood fibers as reinforcing agents. Khazabi *et al.* investigated the effect of the amount of fiber added on the cell morphology of PU foam; the cell structure became irregular as the volume fraction of fibers increased [81]. Gu *et al.* further investigated the effects of fiber on thermal stability and mechanical properties of fiber-reinforced PU foam. The presence of fibers in the strut was observed under scanning electron microscope (SEM) where fibers located in cell struts support foam structure resulting in improved compressive strength of foam [82]. Nonetheless, the presence of fiber is not beneficial for all types of foams. Gu *et al.* found that while the addition of fiber increased the tensile and compressive strengths of semi-rigid foam and the

compressive strength of rigid foam, adding fiber decreased the tensile and compressive strengths of flexible foam and tensile strength of rigid foam [83].

Fillers other than wood fibers have been used to enhance the mechanical properties of foams. Silva *et al.* made PU foam with the reinforcement of cellulose fiber residue. For cellulose fiber residue-modified PU foam, the compressive strength and modulus of foams first decreased and then increased as fiber content increased, but differences were minor [84]. Yuan *et al.* prepared PU foams with the incorporation of wood flour. However, the addition of wood flour caused reduced flexural and tensile strengths because of the poor interface compatibility between wood flour and PU [85]. Zhu *et al.* investigated the effects of cellulose microfibers and nanoclays on the mechanical properties and cell morphology. The addition of microfibers or nanoclays decreased the cell size and increased the compressive strength and modulus decreased after nanoclay content reached a certain amount [87]. There are other fillers with which PU foams are filled: hardwood pulp [88], hay straw [89], paper fiber [90], vegetable fiber [91], glass fiber [92] and carbon fiber [93].

2.3.3 Fiber/Polymer Interface

In composites, the adhesion between fiber and polymer is of great importance; it determines whether the stress can completely transfer to the fiber under loading or stretching and is a crucial factor in terms of the mechanical performance of composites. Poor stress transfer at the fiber/polymer interface can be caused by: (1) poor compatibility between hydrophilic wood fiber and hydrophobic polymer chain, (2) weak interlocking between fiber

and matrix and (3) a large proportion of aggregated and poor dispersed fibers. To address these problems a number of studies have centered on making wood fiber more hydrophobic and the melt blend more stable. The chemical compatibility between fiber and polymer can be improved by using coupling agent. Maleic anhydride-based and silane-based coupling agents are two commonly used ones. These coupling agents are used in preparing PE [94], PP [95], PVC [96], and PS [97] composites. Other chemical modifications used to make fiber more hydrophobic include isocyanate treatment [97], crosslinking, and acetylation [98]. An alkaline treatment is another broadly used approach. The goal of implementing this treatment is to break down the hydrogen bonding between fibers, increase surface roughness of fibers and thus expose more contact area between fiber and matrix [99]. In addition, a partial removal of non-cellulosic components through the alkaline treatment allows cellulose fibrils to rearrange along the direction of fiber axis resulting in a better stress transfer in the fiber [122]. The effects of adding fiber treated by alkaline treatment on mechanical properties of PP composites have been widely investigated. Gwon et al. found that the composite with treated fibers had higher tensile strength compared with that of composite with untreated fibers [100]. Also alkaline treatment is used to provide rough surface of other natural fibers embedded in polymer composites [101].

The interaction between fiber and matrix in the polymeric foams also deserves attention. Coupling agent has been used to modify the fiber-matrix interface. Lee *et al.* used silane-based coupling agent to treat wood fiber [102]. They found that when compared with untreated fiber-reinforced PE foam, treated fiber-reinforced PE foam had fine-celled structure. Bledzki *et al.* studied the effect of coupling agent on mechanical properties of PP foam. It was observed that fiber-reinforced PP foam made from compounds containing

coupling agent showed improvement in cell morphology and tensile and flexural properties of foam [103].

The modification of fiber-matrix interface in PU foam has not been studied extensively. Since PU has a lower hydrophobicity than that of thermoplastic resin due to the presence of carbonyl group, the use of coupling agent may not be an effective way to improve the interface compatibility of fiber-reinforced PU foam. On the other hand, alkaline treatment is considered to be an acceptable route for improvement of the interlocking between fiber and matrix. It is reasonable to assume increase in roughness and surface area of fiber through alkaline treatment can enhance PU foam mechanical properties. In this study, the alkaline treatment of wood fibers was conducted and treated fibers were used for the reinforcement of PU foam.

2.3.4 Role of Catalyst

PU foam synthesis is composed of two chemical reactions: One reaction is for urethane formation, also named as gelling reaction as shown in Figure 2-2. It is known that the formation of the urethane linkages which act as cross-linking joints forms a cross-linking network that is crucial in determining the mechanical performance of the foam. The other reaction is blowing reaction in which isocyanate group reacts with water producing carbon dioxide to expand the volume and fill internal voids (Figure 2-8).

$$O=C=N-R_{(isocyanate)}-N=C=O+2H_2O \rightarrow H_2N-R_{(isocyanate)}-NH_2+CO_{2 (gas)}-NH_2+CO_{2 (gas)}-NH_2+$$

Figure 2-8 Blowing reaction of the synthesis of PU foam

The performance of PU foam is controlled by these two reactions. In the fabrication of PU foam, polyol, isocyanate, and other additives are mixed together. During the mixing, some air bubbles are entrained; these bubbles act as nucleation sites in the foaming process [116]. Once the blowing reaction is initiated, it releases carbon dioxide that expands the bubbles and foam starts rising. As the gelling reaction is initiated, the viscosity of the blend increases, which impedes the foam expansion. Therefore, a good control between blowing and gelling reaction leads to compact cell structure and uniform cell morphology [104].

The rate of blowing and gelling reaction depends on many factors, for instance, reaction temperature, and the type and content of catalyst. Sipaut *et al.* reported that open cell content, cell size and foam density are dependent on the reaction temperature [118]. The catalyst plays a crucial role in achieving a balance between the blowing and gelling reaction and is responsible for achieving a fine cell shape and enhanced mechanical properties [111, 117]. Campanella *et al.* prepared soybean oil-derived PU foams made with and without a catalyst, N, N-Dimethylbenzylamine. It was found that, for the foam made without the catalyst, the blowing reaction dominated over the gelling reaction causing the foam to collapse before curing. The addition of the catalyst allows foam to expand properly. Furthermore, the cellular structure of foam was found to be dependent on the amount of catalyst used [117]. Choe *et al.* studied the effects of the catalyst on kinetic rate of foaming,

density and mechanical properties of foam [105]. The results showed that the cell size decreased as the amount of catalyst increased implying the kinetic rate is governed by the catalyst. However, the amount of catalyst had no significant effects on the density and compressive strength of foam [106].

An amine-based catalyst, compared with organotin and dibutyltin dilaurate (DBTDL), has less toxicity and has been broadly used in industry [107]. Recently, other catalysts such as metal-amine complex have been synthesized and discussed [108]. In this study, foams were prepared using different mono-amines as catalysts and the compressive properties and cell morphology of foams investigated.

2.3.5 Fiber Dispersion

The capability of added fiber to strengthen the cell structure and enhance the foam mechanical behavior depends not only on the stiffness of fibers and the adhesion at the fibermatrix interface, but also on the dispersion of fibers. In a previous study, PP composite was prepared using a mixer for intensive mixing [109]. Mixing is considered as an important step for polymeric foam preparation, especially for the foam containing fibers. Table 2-3 lists previous studies in regard of mixing of polyol-fiber mixture involving different types of natural fibers in the fabrication of PU foam; most of these studies mix the polyol-fiber mixture by hand. However, it is accepted that high degree of fiber dispersion is hard to attain because the viscosity of mixture increases accompanied by the addition of fibers. In this between foam made using hand mixing and that made by mechanical mixing in the cell morphology and mechanical behavior of foam were compared.

Type of fiber	Maximum amount of	Mixing method	Reference
	fibers added (wt%)		
Aspen fiber	1	Hand mixing	[83]
Cellulose fiber	4	Hand mixing	[90] [110]
Cellulose fiber residue	16	Hand mixing	[84]
Cellulose microfiber	1	Hand mixing	[86]
Maple fiber	5	Hand mixing	[88]
Maple fiber	20	Hand mixing	[81] [82]
Birch fiber	10	Mechanical mixing	This work

Table 2-3 PU foams with natural fiber made using different mixing methods

2.4 Problem Statement

Since wood fiber-reinforced PU composite foam has not been reported extensively in the literature, an understanding of the compression performance of the fiber-reinforced foam is required and covered in this study. The adhesion between fiber and polymeric matrix, reagents used in foam preparation, and the dispersion of fiber in the foam are factors controlling foam mechanical performance. In this study, the effect of fiber surface treatment,
catalyst, and mixing method corresponding to each mentioned factor on the compressive performance of PU foam reinforced by fiber was investigated.

2.5 Conclusions

In summary, PU foam reinforced by fiber has been designed as a core material in SIPs. Since there is a growing concern for environmentally friendly production of the fiber-reinforced foam, natural fiber and polyol derived from renewable sources are of major interest.

The adhesion between wood fiber and polymeric matrix affects the mechanical properties of fiber-reinforced foam. It is hypothesized that alkaline treatment can increase fiber coarseness and improve the interlocking between fiber and matrix.

Catalyst and mixing condition are the key manufacturing parameters that affect the cell morphology and mechanical properties of foam. The catalyst controls the balance between gelling and blowing reaction influencing the cell structure of foam and the mixing enhances the dispersion of fibers. Polyol-fiber mixture is mixed by hand in the traditional fabrication of foam. However, for the preparation of foam with high fiber content, mechanical mixing is required due to an increase in the viscosity of the mixture making proper mixing by hand impossible.

3. Experimental Methods

This chapter is composed of two parts. The first part lists the materials used in the fabrication of PU foam and details the process for preparing the foam. The second part discusses how the compression performance of the foam was measured and introduces the characteristic analysis using Fourier transform infrared spectrometer (FTIR) and imaging acquisition of the foam using scanning electron microscopy (SEM), stereo microscopy, and computed X-ray tomography.

3.1 Fiber-reinforced Foam Preparation

3.1.1. Materials

Soybean oil-derived polyols (Soyol ® 2102) were purchased from Urethane Soy Systems (Volga, South Dakota, USA). The viscosity of the polyols is 2181 cps at 25°C and the hydroxyl number is 63 KOH/g according to ASTM-D4274. Polymeric diphenylmethane diisocyanate (PMDI) was obtained from Huntsman Corporation (West Point, Georgia, USA) with a specific gravity of 1.23 g/mL at 25°C, a NCO content of 31.2 % and a functionality of 2.7. Polysiloxane and pyrrolidone used as surfactants and a tertiary amine used as catalyst were donated by Air Products and Chemicals Incorporation (Allentown, Pennsylvania, USA). Ethanolamine was received from Sigma Aldrich (St. Louis, Missouri, USA) and used as catalyst. It was substituted by four other monoamines in the preparation of samples for studying the effect of catalyst: benzyldimethyl amine (CAT-1), 2-(2-aminoethoxy) ethanol (CAT-2), 2-(methylamino) ethanol, and 2-[2-(dimethylamino)ethoxy] ethanol (CAT-4), purchased from Tokyo Chemical Industry American (Portland, Oregon, USA). Deionized water (DI water) was employed as a blowing agent for making foams. Chemi-refiner mechanically pulped medium-density birch wood fibers (*Betula papyrifera* Marsh) were donated by FPInnovations Canada Corporation (Point-Claire, Quebec, Canada). The average length of the fibers is 1.6 mm. These fibers were sieved in 20-30 mesh (American Scale) with fiber lengths ranging between 0.60 and 0.85 mm.

3.1.2. Fiber Treatment

Alkali treatment was used to modify the surface of fibers to improve the interfacial bonding between the fibers and a polymeric matrix. Fibers were dried in an oven at 80°C for 24 hr with a moisture content kept below 5 %. Next, the fibers (10 wt%) were added in NaOH solutions of different concentrations, 2 wt%, 4 wt%, and 7 wt% to assess the effect of NaOH concentration on the fibers. After the NaOH solution treatment, the fibers were rinsed with DI water until a pH value of 7 was reached. Finally, the fibers were once again dried in the oven at 80°C for 24 hr.

Acridine orange dye (AO), purchased from Sigma Aldrich (St. Louis, MO, USA) was employed as a dye to stain fibers for observation under stereo microscope. First, a 6.7×10^{-3} M AO solution in DI water was prepared and then the fibers were immersed in the AO solution at room temperature for 6 hr. The stained fibers were dried in an oven at 80°C for 24 hr. Figure 3-1 (a) and Figure 3-1 (b) show photos and Figure 3-2 (a) and Figure 3-2 (b) show stereo microscopic images of unstained and stained fibers, respectively.



Figure 3-1 Photos of (a) unstained fibers and (b) stained fibers



Figure 3-2 Stereo micrographic images of (a) unstained fibers and (b) stained fibers

3.1.3. Mixing

A mechanical stirrer purchased from Heidolph North American (Elk Grove Village, IL, USA) was used to mix fibers and polyol in the foaming process. The power of the stirrer was 37 W. An impeller, VISCO JET® with the diameter of 60 mm, was purchased from the same company.

3.1.4. Foaming Process

In the section, the preparation of samples was divided into two parts for clarity. The first part described the preparation of samples for the studies regarding the effect of alkaline treatment (Section 4.2) and catalyst (Section 4.3), and the second part for the study associated with the effect of mixing (Section 4.4).

Part I:

A free-rising foaming process was used to make PU foams. In the study of the effect of alkaline treatment, wood fibers were mixed with polyol for 20 min and then certain amounts of surfactants, catalysts, and blowing agent were added into the mixture. Table 3-1 lists the materials used. Then the mixture was mixed for another 5 min. Lastly, PMDI was added into the mixture and hand mixed for 15 s and then the mixture was poured in to a mold allowing the foam to rise and cure. A scheme for the preparation of the foam was shown in Figure 3-3. Foams with various fiber contents (0, 2.5, 5, 7.5, and 10 wt %) of treated or untreated fibers were prepared. In the study of the effect of catalyst, foams were prepared using 100 parts of polyol, 2 parts of pyrrolidone, 3.3 parts of polysiloxane, 3 parts of tertiary amine, 6.7 parts of water and 135 parts of PMDI with different catalysts listed in Table 3-2. Different weights of catalysts were used to keep the same molar ratio with respect to polyol.

Materials		Parts by weight
Soybean oil based polyol	Soyol® 2102	100
Surfactant	Pyrrolidone	2.0
	Polysiloxane	3.3
Catalyst	Tertiary amine	3.0
	Ethanolamine	2.0
Blowing agent	De-ionized water	6.7
Birch wood fibre		0 to 10 wt%
NCO index [*]	PMDI	120

^{*}The NCO index is defined as the ratio of the amount of used isocyanate over the theoretical amount of isocyanate.

Table 3-1Materials used in the preparation of foam



Figure 3-3 Schematic for the preparation of fiber-reinforced foam

Matarials	Parts by weight based on 100 parts of polyol				
Watchais	CAT-0	CAT-1	CAT-2	CAT-3	CAT-4
Ethanolamine	2.0	-	-	-	-
Benzyldimethyl amine	-	3.04	-	-	-
2-(2-Aminoethoxy) ethanol	-	-	2.36	-	-
2-(Methylamion) ethanol	-	-	-	1.68	-
2-[2(Dimethylamion)ethoxy] ethanol	-	-	-	-	3.03

 Table 3-2
 Catalysts used in foam production

Part II:

PU foams were made using a free-rising method. Firstly, fibers were added into polyol and mixed using a mechanical stirrer. The mixture was mixed at designated stirring time and stirring rate. In the study of stirring time, stirring rate was set at 150 rpm and in the

study of stirring rate, stirring time was set at 10 min. Then surfactants, catalysts, and blowing agent were added and mixed for another 5 min. The formulation was shown in Table 3-1. After PMDI was added the mixture was mixed for 20 s. A scheme for the preparation of the foam was shown in Figure 3-4.



Figure 3-4 Schematic for the preparation of fiber-reinforced foam using a mixing stirrer

3.2 Fiber-reinforced Foam Characterization

3.2.1. Mechanical Testing

Specimens (5 cm \times 5 cm \times 3 cm) were cut using a bandsaw. The compressive modulus of the specimen was measured using a Sintech universal testing machine 20 with 1000 lb load cell according to ASTM-D 1621. The specimens were compressed at 15 % deformation and with the rate of crosshead movement set at 2.5 mm/min. A minimum of 5 specimens were tested for each condition, and the mean value and standard deviation recorded.

3.2.2. FQA and FTIR Characterization

The length distribution of wood fibers was measured using a Fiber Quality Analyzer (FQA), purchased from OpTest Equipment Inc. A FTIR TENSOR 27 spectrometer in conjunction with a FTIR attachment (Bruker Optics) was used to record the absorption spectra of wood fibers and PU foams. The sample of the fibers and foams was in pressed KBr pellets. The spectra were obtained over 4000-400 cm⁻¹ at a resolution of 4 cm⁻¹.

3.2.3. Imaging Characterization

Scanning electron microscopy (SEM, Hitachi S-2500), was used to take the micrographs of the surface of wood fibers and the cell morphology of PU foams. The specimens were coated by sputtering a thin layer of Au in vacuum and then examined using SEM. The accelerating voltage was set as 10 kV. To analyze the data and measure the diameter of the fibers, image processing software, Image J in version 1.41 was used.

A high-resolution X-ray tomography (SkyScan 1172) was used to characterize the cell morphology of foams. Foam samples were scanned using a high resolution of 4000 \times 2096 at a 45 kV and a current density 167 μ A with pixel size of 1.5 micron. Software, NRecon developed by SkyScan, was used to reconstruct 2D cross-sectional images of the samples obtained from X-ray tomography.

A 3-D stereo microscope (Leica M165C) equipped with LED based illumination and digital camera was used to image the dispersion of fibers. A sample was cut by using a razor blade. Then the sample was positioned on the center of slide (75 mm \times 25 mm). DI water was used as mounting agent. The slide was finally covered with a clean cover slip (25 mm \times 25 mm).

4. Results and Discussion

4.1 Introduction

This study focuses on the manufacture of wood fiber-reinforced PU foam with enhanced mechanical performance. The goal is addressed in terms of three considerations: (1) fiber/matrix interface, (2) foam preparation and (3) mixing method, and these are factors that play pivotal roles in manufacturing high-performance PU foam.

The interface between wood fibers and a polymeric matrix affects the mechanical properties of PU foam: poor interfacial properties results in a low efficiency of stress transfer from the matrix to fibers under compression loading. The interfacial adhesion between the fibers and the matrix depends on the surface conditions of fibers. In this study, alkaline treatment was used to modify fiber surface in order to improve the interfacial bonding between fibers and the matrix.

The amount and type of catalyst used affect the cell morphology and mechanical properties of the foam. Since a catalytic reaction is sensitive to the chemical structure of the catalyst, in this study, foams were produced using five different mono-amines as catalysts and the characteristics and compression performance of these foams were observed.

Mixing is also a crucial step in the fabrication of foam. For the foam made using a mechanical stirrer, the behaviors of the foam depend on numerous processing factors, including stirring rate and stirring time. In addition, the dispersion of fibers embedded in and

the mechanical performance of the foam made by mechanical mixing were compared with those of the foam made by hand mixing.

4.2 Alkali Treated Fiber-reinforced PU Foams

4.2.1 Characterization of Treated Fiber

The lengths of fibers before and after a treatment with sodium hydroxide (NaOH) solution were measured in order to assess the influence of alkaline treatment on wood fiber. Length distributions of untreated and treated fibers are shown in Figure 4-1 (a) and (b), respectively. Weight mean length of treated fibers was 1.14 mm, similar to that of untreated fibers, 1.15 mm. Fiber curl index of untreated fibers as measured by fiber quality analyzer (FQA) was 0.088 and that of treated fibers was 0.09. It is evident that the alkaline treatment had little impact on the length and curl index of fibers.



Figure 4-1 Distribution of length of (a) untreated and (b) treated fibers measured by FQA

The effect of alkaline treatment on fiber diameter was evaluated by measuring the width of fibers using SEM. The diameter distribution of untreated fibers and that of treated fibers are presented in Figure 4-2. As shown in Figure 4-2, the histogram of treated fiber shifted in the direction of decreasing fiber diameter. The reason for the decrease in fiber diameter is that parts of lignin and other extractives in fiber were removed during the alkaline treatment [2].

The aspect ratio (L/D) of a fiber is defined as the length (L) of the fiber divided by the diameter (D) of the fiber. As mentioned earlier, the length of treated fibers was similar to that of untreated fibers, but the diameter of treated fibers was less than that of untreated fibers. Therefore, treated fibers have higher aspect ratio than that of untreated fibers.

In order to investigate the effect of alkaline treatment on fiber surface, SEM images of untreated fiber and treated fibers were taken. The SEM image of the treated fibers (Figure 4-3 (b)) showed that the surface had become rougher compared with that of the untreated fibers, shown in Figure 4-3 (a); therefore, the surface area able to interact with polymer increased, and better fiber-polymer adhesion may be expected.



Figure 4-2 Distribution of diameters of (a) untreated and (b) treated fibers



Figure 4-3 SEM images of fiber surface of (a) untreated and (b) treated fibers

The effect of different concentrations of NaOH solution used in alkaline treatment on the surface roughness and compressive performance of foam was investigated.

SEM images of fiber surface treated in different concentrations of NaOH solution: 0, 2, 4, and 7 wt% are shown in Figure 4-4; it can be seen that the surface of fibers became rougher as the concentration of NaOH solution increased. However, the number of defects and fractured fibers also increased with the increasing concentration of NaOH solution. This is an expectation at this point that the increase in concentration of NaOH solution reduced the stiffness and strength of the fibers.



Figure 4-4 SEM micrographs of (a) untreated fibers and fibers treated in solutions of (b) 2, (c) 4, and (d) 7 wt% NaOH

Fourier transformation infrared (FTIR) spectroscopy has been used to investigate the components of alkali-treated fibers. Lignocellulosic fiber is mainly composed of cellulose,

hemicelluloses, and lignin. EI-Shekeil *et al.* compared the FTIR spectra of kenaf fibers before and after the alkaline treatment. It was noted that the absorption peak at 1736 cm⁻¹ in the spectrum of treated fibers disappeared attributed to the C=O stretching of hemicelluloses [123]. Cordeiro *et al.* studied the FTIR spectra of alkaline treatment modified eucalyptus, spruce, bagasse and wheat straw. The results indicated that lignin and hemicelluloses were partially removed in the fibers through alkaline treatment [124].

The surface characteristics of fibers before and after alkaline treatment were studied using FTIR spectroscopy. The spectrum of untreated fibers (Figure 4-5 (a)) was compared with that of treated fibers (Figure 4-5 (b)). Peaks observed in both spectra are located at: 3402 cm⁻¹, 2911 cm⁻¹, 1612 cm⁻¹, 1043 cm⁻¹, and 895 cm⁻¹. The formation of an absorbance band at 3402 cm⁻¹ is associated with the C-H stretching vibration of cellulose. An absorbance band at 2911 cm⁻¹ is attributed to the O-H stretching vibration of cellulose. An absorbance band at 1043 cm⁻¹ represents the lateral C-O bonds in cellulose. The presence of these three peaks indicates that cellulose remains in the fibers after alkaline treatment. On the spectra of untreated and treated fibers, the peak at 1612 cm⁻¹ and the peak at 895 cm⁻¹ correspond to the absorbance band of the aromatic vibration and the C=O stretching vibration of lignin and that of the aromatic C-H deformation of lignin, respectively. The results suggested that there was residual lignin in the treated fibers. The spectrum of untreated fibers had two additional characteristic absorbance bands: 1737 cm⁻¹ and 1247 cm⁻¹. The absorbance band at 1737 cm⁻¹ is attributed to the excitation of the C=O stretching of hemicelluloses and the absorbance band centered at 1247 cm⁻¹ results from the excitation of the C-O-C stretching of ether linkages in lignin. It is evident from the presence of the absorbance bands at 1612 cm⁻¹ and

895 cm⁻¹ and absence of the bands at 1737 cm⁻¹ and 1247 cm⁻¹ that the alkaline treatment breaks down lignin molecules.



Figure 4-5 FTIR spectra of (a) untreated and (b) treated fibers

4.2.2 Compression Performance of Treated Fiber-reinforced PU Foam

It was expected that the concentration of NaOH solution with which fibers were treated could affect not only the dimension and surface topography of the fibers but also the compressive strength of the foam reinforced with the fibers. Compressive strengths of foams containing fibers treated in different concentrations of NaOH solution were measured and the results are shown in Figure 4-6. It was observed that the compressive strength of the foam containing fibers treated with 2 wt% NaOH solution was higher than that of the foam containing untreated fibers. However, for the foams containing fibers subjected to 4 and 7 wt% NaOH solutions, the compressive strengths were lower than that of the foam containing fibers treated with the NaOH solution of 2 wt%. The results imply that the alkaline treatment of fiber improves the performance of fiber-reinforced foam; however, high concentration of NaOH can lead to negative outcomes. The improved performance of the fiber-reinforced foam can be attributed to the increased roughness of fiber and the negative outcomes can be attributed to the increased number of defects and fractured fibers observed under SEM (see previous note). Now we have evidence that the fibers were affected and this affects performance.



Figure 4-6 Compressive strengths of foams with the reinforcement of untreated fibers (unWF) and 2.5 wt % of fibers treated with 2 wt% (02WF), 4 wt% (04WF), and 7 wt% (07WF) NaOH solutions

Since the density of foams reinforced with fibers treated by various concentrations of NaOH solutions was not the same and the compressive strength varies as the density of foam, specific strength was calculated to normalize the compressive strength of foam by dividing the compressive strength by foam density and remove the effect of density [86]. Figure 4-7 shows the specific compressive strengths of foam containing the fibers.



Figure 4-7 Specific strengths of foams with the reinforcement of untreated fibers (unWF) and 2.5 wt % of fibers treated with 2 wt% (02WF), 4 wt% (04WF), and 7 wt% (07WF) NaOH solutions

Compressive strengths of foams containing various amounts of untreated and treated fibers were measured and compared in order to estimate the effect of alkali-treated fiber on the mechanical performance of PU foam. The fibers were treated in 2 wt% NaOH solution. The results are shown in Figure 4-8. For untreated fiber-reinforced foam, the compressive strength decreased initially and then increased to a maximum value at a fiber content of 5 wt%. After exceeding the fiber content of 5 wt%, the strength started to decrease with increasing fiber content. For treated fiber-reinforced foam, in the same way, the compressive strength decreased initially, then increased as fiber content increased. Unlike the foam reinforced with untreated fibers, the compressive strength of the foam reinforced with treated

fibers further increased with increasing fiber content. Specific strengths of foams reinforced with various amounts of untreated and treated fibers were computed and are shown in Figure 4-9.



Figure 4-8 Compressive strengths of untreated fiber-reinforced (white) and treated fiberreinforced (grey) foams with various fiber contents from 0 to 10 wt% with an interval of 2.5 wt%



Figure 4-9 Specific compressive strengths of untreated fiber-reinforced (white) and treated fiber-reinforced (grey) foams with various fiber contents from 0 to 10 wt% with an interval of 2.5 wt%

The modulus of fiber-reinforced foam depends not only on the addition of fibers but also on the density of the foam. As fibers act as nucleation sites, foam density changes with the addition of fibers in an uncontrollable way which makes the fiber-reinforced foam a complicated system, and composite theory fails to predict the performance of the foam. Efforts have been made to isolate the effect of each factor. In order to evaluate the reinforcing effect of fiber on modulus, the modulus of pure foam matrix of the appropriate density was computed and compared with the measured modulus of the composite. It has been reported in the literature that the modulus of neat foam (no fiber) varies with foam density in a predictable way. Gibson *et al.* used a cubic member to model the behavior of foam under compression where the modulus of foam is computed based on the linear deflection of beam and derived the equation [13]:

$$\frac{E_o}{E_{PU}} = C_1 \left(\frac{\rho_o}{\rho_{PU}}\right)^2 \tag{1}$$

where E_o is the elastic modulus of foam with no fibers, E_{PU} is the elastic modulus of cell wall material (PU), ρ_o is the density of the foam with no fibers, ρ_{PU} is the density of PU and C_1 is the geometric constant of proportionality.

For the modulus of the matrix of composite foam with the fibers removed, E_x , the Gibson Ashby equation can be applied as below:

$$\frac{E_x}{E_{PU}} = C_1 \left(\frac{\rho_x}{\rho_{PU}}\right)^2 \tag{2}$$

Although E_{PU} may change with the amount of added fibers due to the variation of the degree of crosslinking, it was assumed to remain constant in this case. In addition, the contribution of the fibers to the density of foam is negligible. E_x (fiber content, x = 2.5, 5, 7.5, or 10), was computed by dividing Eq. 2 by Eq. 1 and rearranging to get Eq. 3 as follows:

$$E_x = E_o \times (\frac{\rho_x}{\rho_o})^2 \tag{3}$$

where ρ_x is the density of the matrix for composite foam with the removal of fibers computed as below:

$$\rho_x = \frac{(m_{foam} - m_{fibers})}{V_{foam}}$$
(4)

The difference between the experimental modulus, E_c , and E_x represents the contribution of fiber reinforcement and the ratio was named reinforcing factor (RF):

$$RF = \left(\frac{E_c}{E_x}\right) \tag{5}$$



Figure 4-10 Experimental modulus of untreated $(E_{c(untreated)})$ and treated $(E_{c(treated)})$ fiberreinforced foams and modulus of the matrix for composite foam of untreated $(E_{x(untreated)})$ and treated $(E_{x(treated)})$ fiber-reinforced foams with different fiber contents

Figure 4-10 shows E_c and E_x of foams reinforced with untreated and treated fibers. It can be seen that the difference between E_c and E_x of foam reinforced with untreated fibers

was less than that of foam reinforced with treated fibers. The difference may be related to the nucleation of cells. Eq.1 can be expressed in the form:

$$E_o = C_1 \times \left(\frac{m_{foam}/V_{foam}}{m_{solid}/V_{solid}}\right)^2 \times E_{PU} \cong C_1 \times E_{PU} \times \left(\frac{V_{solid}}{V_{foam}}\right)^2 \tag{6}$$

where m_{foam} and V_{foam} are the mass and volume of foam containing fibers, void, and solid PU, respectively, and m_{solid} and V_{solid} are the mass and volume of cell struts. Assume that m_{foam} equals m_{solid} , E_o is proportionate to the ratio, V_{solid}/V_{foam} , which is the fraction of solid in the foam. Since the surface of treated fiber is rougher, the increased roughness increases the number of nucleation sites and increases the porosity of foam. As the porosity increases, the fraction of solid in the foam decreases and, based on Eq. 7, the computed modulus of foam decreases. This could be the reason that the difference between E_c and E_x of foam reinforced with treated fiber was larger than that of untreated fiber.

Based on Eq. 1, the compressive modulus of foam is proportional to the square of the relative density of foam. Specific modulus was calculated by dividing the compressive modulus of foam by the square of the density of foam. The results of specific modulus are shown in Figure 4-11.



Figure 4-11 Specific modulus of untreated $(E_{sp(untreated)})$ and treated $(E_{sp(treated)})$ fiberreinforced foams with different fiber contents

The relationship between fiber content and RF of foams reinforced with treated and untreated fibers is shown in Figure 4-12. For treated fiber-reinforced foam, there was a great increase in RF as fiber content increased; however, for untreated fiber-reinforced foam, RF increased gradually with increasing fiber content. The results suggested that alkaline treatment improved the fiber/matrix adhesion and increased the contribution of fibers on the compressive performance of foam. This can be attributed to an increase in aspect ratio and surface roughness of treated fibers implying higher degree of stress transfer at the fibermatrix interface contributing to the improvement in the compressive performance of foam reinforced with the fibers.



Figure 4-12 Reinforcing factors of untreated fiber-reinforced and treated fiber-reinforced foams as a function of fiber content

4.2.3 Cell Morphology

An imaging technique, X-ray computed tomography, was used to characterize three dimensional cell structure of fiber-reinforced foam. Figure 4-13 shows the tomographic image of neat foam. As shown in the figure, white area represents void volume of foam and

black lines represent cell edges. Figure 4-14 shows tomographic images of fiber-reinforced foams with different fiber contents. In comparison with the image of foam with no fiber (Figure 4-13), darker parts in Figure 4-14 are considered to be the presence of wood fibers. These parts were located either in cell struts or spread around foam cells; the results support the observations of fibers embedded in foams under SEM reported by Khazabi *et al.* in the literature [81].

The effect of fibers on the cell morphology of foam was examined based on the tomographic results. It was found that the presence of fibers deformed the cell structure of foam. The deformation of the cell structure can be a reason for the decrease in compressive strength observed for foam reinforced with small amount of fibers (from 0 to 5 wt%) as shown in Figure 4-8. Based on the tomographic results, the foam reinforced with 10 wt% of fibers had the largest cell size. Cell size is considered to be a factor that affects the mechanical performance of foam; larger cell size leads to reduced mechanical properties of foam. However, the foam reinforced with 10 wt% of fibers had higher compressive strength and modulus than other foams with lower fiber contents. This is because the increase in the interfacial adhesion between fibers and a polymer matrix compensates for the undesirable effect of larger cell size. It is clear that the interaction between fibers and the polymer matrix plays a key role in enhancing the mechanical performance of fiber-reinforced foam.



Figure 4-13 Reconstructed X-ray tomographic image of neat foam





Figure 4-14 Reconstructed X-ray tomographic images of foams with (a) 2.5, (b) 5, (c) 7.5, and (d) 10 wt% treated fibers

4.2.4 Conclusions

Wood fibers were treated with NaOH solution to improve the interfacial bonding between fibers and a polymeric matrix. The concentration of the NaOH solution affects the surface of the fibers and further the compression performance of the foam reinforced with the fibers. Reinforcing factor was introduced to assess the reinforcing capability of the treated and untreated fibers embedded in the foam. It was concluded that an appropriate alkaline treatment can enhance the reinforcing capability of fibers and the compression performance of the foam reinforced with the fibers.

4.3 Effect of Catalyst on the Compression Performance of Foams

Foams were made using various mono-amines as catalysts and the compressive strength and cell morphology of the foams were measured in order to investigate the effect of the catalyst.

The preparation of PU foam consists of two reactions: blowing and gelling reaction. The catalyst plays a crucial role in achieving a balance between the blowing and gelling reaction and is responsible for achieving a fine cell shape and enhanced mechanical properties [111, 117]. In this study, five mono-amines were used as catalysts in the fabrication of PU foams. The chemical structures of these catalysts are displayed in Figure 4-15. Foams were made using each catalyst with other reagents (blowing agent and surfactants) and the kinetic rate of foam formation and the compression performance of the foams were measured.

4.3.1 Reaction Profile of PU Foams

The kinetic rate of reaction was estimated by measuring cream time, gel time and rise time during foam-forming process. Cream time is defined as the time counted from pouring a mixture into a mold to the initial point of rising at which the mixture is saturated with blowing gas. Gel time is the time counted from the same starting point to the formation of thin polymer string at which the cross-linking reaction between urethane and urea takes place. Rise time is the time for the mixture to reach the maximum height. Cream time is related to the rate of blowing reaction while gel time associated with the rate of gelling reaction [106]. By comparing these kinetic times, the catalyst that favors blowing reaction or gelling reaction can be determined.

Kinetic times of foams made with different types of catalysts are listed in Table 4-1 for the assessment of the dynamics of foaming reactions. According to Table 4-1, foams made with catalysts, CAT-0, CAT-1, and CAT-4 had shorter cream time compared with those made with, CAT-2, and CAT-3. The results implied that these catalysts (CAT-0, CAT-1, and CAT-4) have relatively higher catalytic activity for the blowing reaction than other catalysts (CAT-2, and CAT-3).



Figure 4-15 Chemical structures of (a) ethanolamine (CAT-0), (b) benzyldimethyl amine (CAT-1), (c) 2-(2-aminoethoxy) ethanol (CAT-2), (d) 2-(methylamino) ethanol (CAT-3), and (e) 2-[2-(dimethylamino)ethoxy] ethanol (CAT-4)

The relationship between rise height and time was examined in order to estimate the effect of catalyst on the kinetic rate of reaction in the preparation of foam. Figure 4-16 shows rising profiles of foams made using various catalysts. The results showed that catalyst type affected the slope of rise height versus time at the beginning of foaming. The slope represents the rising rate of foam. In Figure 4-16, there was a trend that foams made using catalysts, CAT-0, CAT-1, CAT-4 had a higher rising rate. This indicates that these catalysts have higher catalytic activity for the blowing reaction than CAT-2 and CAT-3.

Sample	Cream time (min:s)	Gel time (min:s)	Rise time (min:s)	
CAT-0	0:32	0:37	1:31	
CAT-1	0:37	0:52	1:42	
CAT-2	0:51	1:02	2:38	
CAT-3	0:45	1:12	2:16	
CAT-4	0:22	0:33	0:54	

Table 4-1Cream time, gel time and rise time of PU foam prepared with catalyst: CAT-0,
CAT-1, CAT-2, CAT-3, and CAT-4



Figure 4-16 Rise height of PU foams prepared with catalyst: (a) CAT-0, (b) CAT-1, (c) CAT-2, (d) CAT-3, or (e) CAT-4 as a function of time

4.3.2 Effect of Catalyst on the Compression Behavior of Foam

The effect of catalysts (CAT-0, CAT-1, CAT-2, CAT-3, and CAT-4) on the compression performance of corresponding foams was investigated. In the literature, Choe *et al.* reported that the content of catalyst had an effect on the cell size, kinetic rate, and compressive strength of PU foam [105]. The compressive strength of the foam made using a physical blowing agent increased as the content of blowing catalyst increased due to the collapsed of foam cells while it was independent of the content of gelling catalyst. However,

for the foam made using water as a blowing agent, the compressive strength was independent of the content of blowing and gelling catalysts [105]. The compressive strength and modulus of the foams made using various catalysts are shown in Figure 4-17 and Figure 4-18, respectively. In Figure 4-17, it can be seen that the foam made using CAT-1 had highest compressive strength. Similar results were observed in the results of compressive modulus while the foam made with CAT-3 had comparable value with that of the foam made using CAT-1 as shown in Figure 4-18.



Figure 4-17 Measured compressive strengths of PU foams made with CAT-0, CAT-1, CAT-

2, CAT-3, and CAT-4


Figure 4-18 Measured compressive moduli of PU foams made with CAT-0, CAT-1, CAT-2, CAT-3, and CAT-4

The catalyst affects the density of foam and has effect on the results shown in Figure 4-17 and Figure 4-18. Densities of foams made with various catalysts were measured and are shown in Figure 4-19. As shown in Figure 4-19, the density of foam depended on the type of catalyst with which foams were made.



Figure 4-19 Densities of PU foams produced with various catalysts (CAT-0, CAT-1, CAT-2, CAT-3, and CAT-4)

Hence, in order to isolate the effect of catalyst on the compression performance of foam from any density changes, the compressive modulus of each sample was divided by the square of corresponding density (See Section 4.2.2). The calculated value was named as specific modulus as shown in Figure 4-20. From Figure 4-20, it is apparent that foams made using catalysts, CAT-0, CAT-2 and CAT-4 shared similar specific moduli while the specific modulus of foam made using CAT-3 was larger than that of foam made using CAT-1. The specific modulus results cannot be explained by considering the catalytic activity of the catalysts, toward blowing reaction or gelling reaction, since the materials formed by the

blowing reaction and gelling reaction are urea and urethane, respectively. This is because urea and urethane can extend the cross-linking networks and increase the stiffness of polymer.



Figure 4-20 Specific compressive moduli of PU foams produced with catalysts: CAT-0, CAT-1, CAT-2, CAT-3, and CAT-4

An activation mechanism of catalyst in the gelling reaction proposed by Maris *et al.* [112] was used to interpret the results of specific modulus of foams made using catalysts with similar chemical structures but different chain lengths, CAT-2 and CAT-3. A schematic diagram of the mechanism is shown in Figure 4-21. The hydroxyl group on the polyol reacts with the catalyzed isocyanate group and forms an intermediate which then turns into a urethane linkage which connects polymer chains. As more urethane linkages are formed, the cross-linking degree and stiffness of polymer network increases. Based on the chemical

structure of catalysts, it was hypothesized that CAT-3, with shorter chain length and smaller molecular size than CAT-2, has less steric hindrance which facilitates the formation of urethane linkages. Hence, it was expected that the specific modulus of foam made with CAT-3 is higher than that of foam made with CAT-2. However, the results of specific modulus (Figure 4-20) demonstrated that foams made using CAT-2 and CAT-3 had similar values. It was concluded that the steric hindrance of the chemical structure of catalyst is a factor that does not affect the formation of urethane linkages to a statistically significant extent.



Figure 4-21 Schematic diagram of the activation mechanism of isocyanate-hydroxyl gelling reaction

4.3.3 FTIR Characterization of PU Foams

FTIR spectra of foams made with the different catalysts were measured to analyze the influence of the catalysts on functional groups in the foams. The FTIR spectra of foams are

displayed in Figure 4-22. From Figure 4-22, foams made with all catalysts shared similar FTIR patterns. A broad absorption peak at 3320 cm⁻¹ is attributed to the N-H stretching vibration of urethane linkages connected with hydrogen bonds [113]. An absorption peak around 2272 cm⁻¹ represents the stretching of the NCO group of isocyanate which indicates that the incomplete conversion of isocyanate. Absorption peaks at 1224 cm⁻¹ and 1513 cm⁻¹ indicate the presence of N-H and N-C bond in urethane linkages, respectively. The absorption peak at 1725 cm⁻¹ associated with the C=O stretching of urethane linkages [114] was observed in the spectra of foams made with CAT-0 and CAT-1, while the peak was hardly seen in the spectra of other foams. However, the reason for this has not yet been determined.



Figure 4-22 FTIR spectra of PU foams prepared with CAT-0, CAT-1, CAT-2, CAT-3, and CAT-4

FTIR spectra were used to estimate the concentration distribution of urethane and urea linkages of foam. The spectra were characterized using a curve-fitting analysis. Figure 4-23 shows the curve fitting results for the foam made using CAT-0. As shown in the Figure 4-23, the peaks can be determined by fitting the Gaussian distribution with the spectrum. Absorbance peaks around 1670 cm⁻¹, 1710 cm⁻¹, and 1730 cm⁻¹ were observed. The peak at 1670 cm⁻¹ is attributed to the carbonyl group of urea linkages. The absorbance peak at 1710 cm⁻¹ and 1730 cm⁻¹ indicates the presence of hydrogen bonded carbonyl group of urethane

linkages and free urethane linkages, respectively. Table 4-2 shows the relative concentration of urea and urethane linkages of the foam made using catalysts (CAT-0, CAT-1, CAT-2, CAT-3, and CAT-4). It can be seen that the foam made using CAT-0, CAT-1, and CAT-4 had lower relative concentration of urea linkages than CAT-2, and CAT-3. Based on the kinetic time and rise height results, it was expected that the concentration of the urea linkages of the foam made using CAT-0, CAT-1, and CAT-4 would be higher than those of CAT-2, and CAT-3 due to the higher catalytic activity toward blowing reaction. However, the opposite results were observed in Table 4-2. Since the catalytic system for foaming is complex, there are factors in addition to the activity of catalyst that have an effect on the proportion of urethane and urea linkages formed in the foam. A more complete understanding is required to further investigate the catalytic system for making PU foam and the composition of the foam.



Figure 4-23 Curve-fitted FTIR spectrum of the foam made using CAT-0 as a catalyst in the $1350-1800 \text{ cm}^{-1}$ region

Relative concentration (%)					
	CAT-0	CAT-1	CAT-2	CAT-3	CAT-4
Urethane linkage	80.0	81.6	69.6	61.7	85.6
Free Urethane	44.8	42.8	47.9	37.8	48.6
H-bond Urethane	35.2	38.8	21.7	23.9	37.0
Urea linkage	20.0	18.4	30.4	38.3	14.4

Table 4-2Relative concentration of urea and urethane linkages of the foams made using
catalysts, CAT-0, CAT-1, CAT-2, CAT-3, and CAT-4

4.3.4 Morphology of PU Foams

The mechanical properties of PU foam depend on the cell structure and cell distribution of foam. The presence of defects in the cell structure leads to poor mechanical performance of foam. It has been demonstrated that the cell structure is an important factor associated with the compressive strength of PU foam [115]. This section deals with the effect of different catalysts on the cell morphology.

Figure 4-24 shows SEM images of PU foams made using different catalysts. It can be seen that foams made using catalysts with the exception of CAT-3 had fine cell structure

with foam cells in polygon shapes. The cell morphology of foams was similar to that obtained in the literature [86]. Irregular cell shape and uneven cell size distribution were observed in the foam prepared with CAT-3. In view of the cell morphology of foam, CAT-3 is less suitable for the preparation of PU foam, however, the selection of catalyst cannot be made based on the results of the compression performance or cell morphology of foam results alone.





Figure 4-24 SEM micrographs of PU foams made with (a) CAT-0, (b) CAT-1, (c) CAT-2, (d) CAT-3, and (e) CAT-4 in the foam rising direction

4.3.5 Conclusions

In this section, foams were made using five different monoamines as catalysts in order to study their effect on the compression performance and cell morphology of the foams. The catalytic activity of amine catalysts was assessed by the kinetic time and rising profile results. The influence of the catalysts on functional groups in the foams was investigated by taking FTIR spectra. Based on the compression results, it was shown that foams made using different catalysts shared similar specific compressive moduli. An activation mechanism was used to interpret these results indicating that the steric hindrance of the chemical structure of catalyst is a factor that does not affect the compression performance of the foam to a statistically significant extent. From the SEM images of the foams, the cell morphology of the foams was similar except for the foam made using the catalyst, CAT-3.

4.4 Characterization of Foams Made Using Different Mixing Methods

In fiber-reinforced polymer composite, fibers make polymer matrix mechanically stronger as the fibers are well dispersed. It is considered that the same is true in the fiberreinforced foam system. In order to disperse the fibers uniformly in the PU matrix, a mechanical stirrer was used for mixing.

4.4.1 Effect of Stirring Time on the Compression Response of Foam

In order to investigate the effect of mixing on the mechanical properties of foam, foams were made under various mixing conditions and the effect of mixing conditions on the mechanical performance of foam was measured. Specified amounts of fiber and polyol were blended using either hand mixing or a mechanical stirrer. It was hypothesized that a truly thorough mixing can be achieved using the agitation force provided by the mechanical stirrer especially for longer stirring times and enhanced mechanical performance of foam can be achieved.

Stirring time and stirring rate are two important variables involved in the preparation of foam. In this section, the effect of stirring time on the compression performance of foam was examined. Foams made using the same formulation but under various stirring times were prepared and the compressive strength and modulus of foams were measured. The results are shown in Figure 4-25 and 4-26. Foams made with different stirring times shared the similar strength and modulus results except the foam mixed for 30 minutes, which had a decrease in both strength and modulus results.



Figure 4-25 Compressive strength of fiber-reinforced foam as a function of stirring time



Figure 4-26 Compressive modulus of fiber-reinforced foam as a function of stirring time

RFs of foams made with different stirring times were computed and are shown in Figure 4-27 to isolate the effect of the fibers on the results. The results showed that no change of RF over stirring time was evident. This implies that the stirring time of 10 min was enough to disperse the fibers evenly. It was observed that the foams made using stirring time less than 10 min had large pores and collapsed easily.



Figure 4-27 Reinforcing factor of fiber-reinforced foam as a function of stirring time

In order to examine the effect of stirring time on the cell morphology of foam, SEM images were acquired and are shown in Figure 4-28. As SEM image of foam made with 10 min stirring time (Figure 4-28 (a)) and that of foam made with 30 min stirring time (Figure 4-28 (b)) were compared, no significant difference was observed in terms of the cell morphology of foam. In addition, tomographic images were obtained in order to further study the effect of stirring time on the cell morphology of foam and are shown in Figure 4-29. It can be seen that the cell size of the foam made with 30 min stirring time (Figure 4-29 (b)) was a bit larger than that of the foam made with 10 min stirring time (Figure 4-29 (a)). Although SEM imaging is not a direct way of evaluating the degree of fiber dispersion, fiber aggregates and significant changes in the cellular structure observed under SEM may provide

an evidence of poor dispersion of fibers. The results suggested that the increase in stirring time had no significant effect on the cell morphology of the foam since the fibers were already well dispersed after 10 min stirring.



Figure 4-28 SEM micrographs of foams made with (a) 10 and (b) 30 min stirring times



Figure 4-29 Reconstructed X-ray tomographic images of foams made with (a) 10 and (b) 30 min stirring times

4.4.2 Effect of Stirring Rate on the Compression Response of Foam

Section 4.4.1 discussed the effect of stirring time on the compression behavior and cell morphology of fiber-reinforced foam. This section sheds light on the effect of stirring rate on the above-mentioned properties. Foams were prepared with different stirring rates but at the same stirring time, 10 min, and the compression response of foams was measured in order to investigate the effect of stirring rate on the compression performance of foam. Figure 4-30 and Figure 4-31 show the compressive strength and modulus of foams made with different stirring rates, respectively. It can be noted that the strength and modulus of foam decreased as the stirring rate increased.

RFs of foams prepared with varying stirring rates were computed in order to isolate the effect of fiber and the results are shown in Figure 4-32. From Figure 4-32, it can be seen that RF decreased with increasing stirring rate. The decrease in RF can be attributed to the breakage of fibers during mixing. High stirring rate and force cause an increased number of fiber fractions and reduce the reinforcing capability of the fibers.



Figure 4-30 Compressive strength of fiber-reinforced foam as a function of stirring time



Figure 4-31 Compressive modulus of fiber-reinforced foam as a function of stirring rate



Figure 4-32 Reinforcing factor of fiber-reinforced foam as a function of stirring rate

SEM and tomographic images of foams made with various stirring rates were obtained in order to evaluate the effect of stirring rate on the cell morphology of foam. Figure 4-33 shows the SEM image of foam made with 600 rpm stirring rate. Compared with the SEM image of foam made with 150 rpm stirring rate (Figure 4-28 (a)), the foam made with 600 rpm stirring rate had irregular cell shapes and larger cell size. Figure 4-34 presents the tomographic image of the foam made with 600 rpm stirring rate. In Figure 4-34, cell edge of foam became less clear compared with that of foam made with 150 rpm stirring rate (Figure 4-29). This is due to a decreased amount of unbroken fibers. High stirring rate makes it possible to break down fibers into fractions. Since both the number of unbroken fibers spread over cells or collected in cell struts reduced, the cell edge of foam became less clear. It can be noted that high stirring rate in the preparation of foam led to increased amount of fractured fibers embedded in the foam.



Figure 4-33 SEM micrographs of foam made with 600 rpm stirring rate



Figure 4-34 Reconstructed X-ray tomographic image of foam made with 600 rpm stirring

rate

4.4.3 Mixing Method Comparison on the Compression Response of Foam

Foams were prepared using hand and mechanical mixing and their compressive strength and modulus were measured in order to examine the difference between the foam made by hand mixing and that made by mechanical stirring. For the foam prepared using the mechanical mixing, the stirring time was chosen to be 10 min, enough for good dispersion of fibers, and the stirring rate was 150 rpm, where fibers had highest reinforcing capability. The compressive strength of foams made using hand mixing and mechanical stirring is shown in Figure 4-35 and the modulus of these foams is shown in Figure 4-36. Figure 4-35 reveals that the compressive strengths of foams fabricated using hand mixing followed the same trend as those of foam using a mechanical stirrer for fiber contents less than 5 wt%. Above 5 wt%, there was a difference in compressive strength between the foam made by hand mixing and that made by mechanical mixing. Similar trends were seen in compressive modulus results of foams made using different mixing methods as shown in Figure 4-36.



Figure 4-35 Compressive strengths of foams made using hand (white) and mechanical (grey)







RFs of foams made using hand mixing and a mechanical stirrer were computed in order to evaluate the reinforcing capability of fibers. The results are shown in Figure 4-37. From Figure 4-37, it can be seen that the reinforcing capability of fibers embedded in the foam made using the mechanical stirrer decreased as fiber content increased while that of fibers in the foam made using hand mixing increased with increasing fiber content. This is because the stirring force breaks fibers into fractions during mixing. Thus broken fibers have reduced reinforcing capability and contribute less to the compressive strength and modulus of foam. Especially in the preparation of foam with high fiber content, this phenomenon became more significant as the viscosity of blend increased.



Figure 4-37 Reinforcing factors of foams made using hand and mechanical mixing as functions of fiber content

Morphological analysis techniques, SEM and tomography, were used to obtain the morphology of foams made using different mixing methods. SEM and tomographic images of foams made using hand mixing and a mechanical stirrer were recorded. Figure 4-38 shows SEM images of the foams made using hand mixing and the mechanical stirrer. It was observed that the cell morphology of the foam made using hand mixing was similar to that of the foam made using mechanical mixing. Tomographic images of the foam made using hand mixing and the foam made using mechanical stirrer were obtained and are shown in Figure 4-39. Tomographic results exhibited that fiber clusters were formed and observed in the foam made using hand mixing (Figure 4-39 (a)). The formation of fiber clusters indicated that fibers were less dispersed. The degree of fiber dispersion affects the cell morphology and mechanical performance of foam. The fibers well dispersed in the polymer matrix were assumed to have higher reinforcing capability. However, the results showed that the RF of fibers in the foam made using mechanical mixing decreased as the fiber content increased (Figure 4-37). It was concluded that the effect of fiber breakage outweighs the effect of fiber dispersion.



Figure 4-38 SEM micrographs of foams containing 10 wt% of fibers made using (a) hand and (b) mechanical mixing



Figure 4-39 Reconstructed X-ray tomographic images of foam containing 10 wt% of fibers made using (a) hand and (b) mechanical mixing

The dispersion of fibers embedded in the foams made using hand and mechanical mixing was evaluated using a stereo microscope. Figure 4-40 shows microscopic images of foams made using hand mixing and mechanical mixing. It is evident that fiber clusters were observed in the foam made using hand mixing but not in the foam made using a mechanical stirrer. The results supported the argument that fibers tended to form fiber clusters in the foam made using hand mixing, consistent with the results observed using X-ray computed tomography.



Figure 4-40 Stereo microscopic images of foams containing 5 wt% of fibers made using (a) hand and (b) mechanical mixing

4.4.4 Conclusions

Based on the investigated effect of mixing methods on the compression performance and cell morphology of fiber-reinforced foam, it can be concluded that, in the foam made using a mechanical stirrer, fibers were well dispersed but the reinforcing capability of the fibers was reduced. The reinforcing capability of the fibers in the foam made using the mechanical stirrer decreased as fiber content increased attributed to the breakdown of fibers during mixing, while the reinforcing capability of fibers in the foam made using hand mixing increased with increasing fiber content. Fiber clusters existed in the foam made using hand mixing but not in the foam made using mechanical mixing. In summary, the use of the mechanical stirrer provides a reproducible and controllable way for foam preparation and makes it possible to spread fibers evenly while the high shearing force of mechanical agitation breaks down the fibers and reduces the reinforcing ability of the fibers.

5. Conclusions and Recommendations

5.1 Conclusions

In this study, the effects of alkaline treatment, catalyst and mixing method on the cell morphology and compression behavior of wood fiber-reinforced PU foam were investigated. Conclusions that can be drawn from the experimental results are presented below:

- 1. Alkaline treatment improved the adhesion between fibers and polymeric matrix and increased the reinforcing capability of fibers and the compression performance of the foam reinforced with the fibers.
- 2. After the alkaline treatment, the length of fibers remains unchanged while the diameter of fibers decreased due to the removal of other extractives. A comparison of FTIR absorbance spectra of treated fiber and untreated fiber confirmed that the breakage of ether linkage in lignin.
- 3. The compression performance of treated fiber-reinforced foams depended on the concentration of NaOH solution in which fibers were processed. High concentrations of NaOH solution resulted in a rougher surface of fiber causing better adhesion at the fiber-matrix interface. However, the number of fractured fibers increases with increasing NaOH concentration, reducing the stiffness of fibers.
- 4. The effect of using different monoamines as catalysts on the cell morphology and compression behavior of foam was studied. It was observed that the type of the

catalyst has no significant effect on the compression performance of foam. SEM images showed that foams made using various catalysts have similar morphologies with the exception of foam made using catalyst, 2-(methylamino) ethanol.

- 5. The effects of stirring time and stirring rate on the compression performance of foam were investigated. It can be seen that the reinforcing capability of fibers embedded in foam is independent of the change in stirring time, indicating that 10 min stirring time was enough for good dispersion of fibers, while the reinforcing capability decreases with increasing stirring rate, due to increased number of fiber fractions.
- 6. As the foam made using hand mixing was compared with the foam made using a mechanical stirrer, it was observed that the reinforcing capability of fibers embedded in the foam made using the mechanical stirrer decreased as fiber content increased while the reinforcing capability of fibers in the foam made by hand mixing increased with increasing fiber content. About fiber dispersion, fibers in the foam made by mechanical mixing had lower chance of forming fiber clusters compared with fibers in the foam made by hand mixing.

5.2 Recommendations

This study reported that the compressive properties of wood fiber-reinforced PU foam can be enhanced through alkaline treatment of fibers and investigated the effect of catalyst and mixing method on the cell morphology and compression behavior of the foam. It is suggested that the use of alkali treatment can be extended to other natural fibers added in PU foams for reinforcing purpose. In the fabrication of PU foam, the effect of other amines used as catalysts on the mechanical properties of the foam should be studied. Moreover, further study on the type of propeller blade used in blending fiber-polyol mixture and the modulus development of the fiber-reinforced foam is recommended. In general, more efforts devoted to the exploration of the factors affecting the preparation of the wood fiber-reinforced foam and the way to enhance the mechanical properties of the foam are necessary.

6. Reference

- [1] Gibson, L. J., "The hierarchical structure and mechanics of plant materials", Journal of the Royal Society Interface, doi: 10.1098/rsif.2012.0341, 2012
- [2] Peng, X., Zhong, L., Ren, J., Sun, R., "Laccase and alkali treatments of cellulose fibre: Surface ligning and its influences on fibre surface properties and interfacial behaviour of sisal fibre/phenolic resin composites", Composites Part A: Applied Science and Manufacturing, Vol. 41, No. 12, Pg. 1848, 2010
- [3] Schidrowitz, P. and Goldsbrough, H. A., British Patent No. 1,111, 1914
- [4] Ruskin, B. L., "Handbook of Foamed Plastics", Lake Publishing, Libertyville, Ill., 1965
- [5] Munters, C. G. and Tandberg, J. G., U.S. Patent No. 2,023,204, 1935
- [6] Bayer, O., "Das Di-isocyanat-Polyadditionsverfahren (Polyurethane)", Angewandte Chemie, Vol. 59, No. 9, Pg. 257-272, 1947
- [7] Simon, E. and Thomas, F. W., U.S. Patent No. 2,577,279, 1945
- [8] Simon, E. and Thomas, F. W., U.S. Patent No. 2,577,280, 1945
- [9] Simon, E. and Thomas, F. W., U.S. Patent No. 2,577,281, 1945
- [10] Buist, J. M. and Lowe, A., Transaction and Journal of the Plastics Institute, Vol. 27, Pg. 67, 1959
- [11] Frisch, K. C. and Saunders, J. H., "Plastic Foams, Part I", Marcel Dekker, New York, United States, 1972
- [12] Lee, S. T. and Ramesh, N. S., "Polymeric Foams: Mechanisms and Materials", CRC Press, Florida, United States, 2004
- [13] Gibson, L. J. and Ashby, M. F., "Cellular Solids: Structure and Properties", University of Cambridge, Cambridge, United Kingdom, 1997
- [14] Rodríguez Pérez, M. A., "Crosslinked polyolefin foams: production, structure, properties, and applications", Advances in Polymer Science, Vol. 184, Pg. 97, 2005
- [15] Eaves, D., "Polymer Foams Trends in Use and Technology", Rapra Technology Ltd., Shawbury, United Kingdom, 2001
- [16] Naguib, H. E., Park, C. B., Panzer, U., and Reichelt, N., "Strategies for achieving ultra low-density polypropylene foams", Polymer Engineering and Science, Vol. 42, No. 7, Pg. 1481, 2002
- [17] Yoshimura, S. and Kuwabara, H., U.S. Patent No. 4,464,484, 1984
- [18] Yu, L., Zhu, Q., and Yu, J., "Development and application of expanded polypropylene foam", Journal of Wuhan University of Technology-Mater. Sci. Ed., Vol. 28, No. 2, Pg. 373, 2013
- [19] Imielińska, K., Guillaumat, L., Wojtyra, R., Castaings, M., "Effects of manufacturing and face/core bonding on impact damage in glass/polyest-PVC foam core sandwich panels", Composites Part B: Engineering, Vol. 39, No. 6, Pg. 1034, 2008
- [20] Koçhan, C., Neşer, G., and Özes, Ç., "The fracture toughness of PVC/GRP sandwich systems in marine environment", Advanced Composites Letters, Vol. 17, No. 1, Pg. 23, 2008
- [21] Winterling, H. and Sonntag, N., "Rigid polystyrene foam (EPS, XPS)", Kunststoffe, Vol. 10, Pg. 32, 2011
- [22] Lanzani, F. and Mauri, R., European Patent No. 0849309
- [23] Park, C. P., and Chaudhary, B. I., and Imeokparia, D. D., U.S. Patent No. 6,093,752

- [24] Suh, K. W., Park, C. P., Maurer, M. I., Tusim, M. H., Genova, R. De, Broos, R., and Sophiea, D. P., "Lightweight cellular plastics", Advanced Materials, Vol. 12, No. 23, Pg. 1779, 2000
- [25] Omer, S. A., Riffat, S. B., and Qiu, G., "Thermal insulations for hot water cylinders: A review and a conceptual evaluation", Building Services Engineering Research and Technology, Vol. 28, No. 3, Pg. 275, 2007
- [26] Ashida, K., Saiki, K., Goto, J., and Sasaki, K., "Polyisocyanurate foams modified by thermally stable linkages", ACS Symposium Series, Vol. 669, Pg. 81, 1997
- [27] Moriya, K., Tani, K., and Tanaka, K., "Novel fire resistant high index polyisocyanurate foam for exterior finish insulation application", Proceeding of 33rd Annual Polyurethane Technical/Marketing Conference, 1990
- [28] Moriya, K., Naruse, A., Kurita, M., and Sasaki, K., "Fire resistant all water blown polyisocyanurate foam", Proceeding of the Polyurethane Expo, 1996
- [29] "2013 Plastics Research Review" (report code: PLS051D), BCC Research, Massachusetts, United States, 2013
- [30] Walsh, K., "Non-plastic foams drive demand", Chemical Week, Vol. 165, No. 32, Pg. 40, 2003
- [31] Allcock, H. R., Lampe, F. W., and Mark, J. E., "Contemporary polymer chemsitry", Pearson Education Inc., New Jersey, United States, 2003
- [32] Dušek, K., Špirková, M., Havlíček, I., "Network formation of polyurethanes due to side reactions", Macromolecules, Vol. 23, No. 6, Pg. 1774, 1990
- [33] Schmelzer, H. G., "Polyurethanes for flexible surface coatings and adhesives", Journal of Coated Fabrics, Vol. 17, No. 3, Pg. 167, 1988
- [34] Engels, H. W., Pirkl, H. G., Albers, R., Albach, R. W., Krause, J., Hoffmann, A., Casselmann, H., and Dormish, J., "Polyurethanes: versatile materials and sustainable problem solvers for today's challenges", Angewandte Chemie International Edition, Vol. 52, No. 36, Pg. 9422, 2013
- [35] Kanyanta, V., and Ivankovic, A., "Mechanical characterisation of polyurethane elastomer for biomedical applications", Journal of the Mechanical Behavior of Biomedical Materials, Vol. 3, No. 1, Pg. 51, 2010
- [36] Tan, S., Abraham, T., Ference, D., Macosko, C. W., "Rigid polyurethane foams from a soybean oil-based polyol", Polymer, Vol. 52, No. 13, Pg. 2840, 2011
- [37] Lee, S. T., Park, C.B., and Ramesh, N. S., "Polymeric foams: technology and science of polymeric foams", CRC Press, Florida, United States, 2007
- [38] Szycber, M., "Handbook of polyurethanes", CRC Press, Florida, United States, 2000
- [39] Narine, S. S., Kong, X., Bouzidi, L., Sporns, P., "Physical properties of polyurethanes produced from polyols from seed oils: II. Foams", Journal of the American Oil Chemists' Society, Vol. 84, No. 1, Pg. 65, 2007
- [40] Baser, S. A., and Khakhar, D. V., "Castor oil-glycerol blends as polyols for rigid polyurethane foams", Cellular Polymers, Vol. 12, No. 5, Pg. 340, 1993
- [41] Thames, S. F., Bautista, M. O., Yu, H., Panjnani, K. G., Wang, M. D., "Lesquerella: renewable resource for industrial coatings and polyurethane foams", Proceeding of International SAMPE Symposium and Exhibition, 1996
- [42] Hu, Y. H., Gao, Y., Wang, D. N., Hu, C. P., Zu, S. Vanoverloop, L., and Randall D., "Rigid polyurethane foam prepared from a rape seed oil based polyol", Journal of Applied Polymer Science, Vol. 84, No. 3, Pg. 591, 2002

- [43] Chian, K. S., and Gan, L. H., "Development of a rigid polyurethane foam from palm oil", Journal of Applied Polymer Science, Vol. 68, No. 3, Pg. 509, 1998
- [44] Beltrán, A. A., and Boyacá, L. A., "Production of rigid polyurethane foams from soybased polyols", Latin American Applied Research, Vol. 41, No. 1, Pg. 75, 2011
- [45] Mishra, D., and Sinha, V. K., "Value-based polymer precursors from paper waste and its application in polyurethane foams", Journal of Cellular Plastics, Vol. 46, No. 1, Pg. 15, 2010
- [46] Sandridge, R. L., U.S. Patent No. 3,211,764, 1965
- [47] Yan, Y., Pang, H., Yang, X., Zhang, R., and Liao, B., "Preparation and characterization of water-blown polyurethane foams from liquefied cornstalk polyol", Journal of Applied Polymer Science, Vol. 110, No. 2, Pg. 1099, 2008
- [48] Gao, L. L., Liu, Y. H., Lei, H., Peng, H., and Ruan, R., "Preparation of semirigid polyurethane foam with liquefied bamboo residues", Journal of Applied Polymer Science, Vol. 116, No. 3, Pg. 1694, 2010
- [49] Abdel Hakim, A. A., Nassar, M., Emam, A., and Sultan, M., "Preparation and characterization of rigid polyurethane foam prepared from sugar-cane bagasse polyol", Materials Chemistry and Physics, Vol. 129, No. 1-2, Pg. 301, 2011
- [50] Zhao, Y., Yan, N., and Feng, M., "Polyurethane foams derived from liquefied mountain pine beetle-infested barks", Journal of Applied Polymer Science, Vol. 123, No. 5, Pg. 2849, 2012
- [51] Yao, Y. G., Yoshioka, M., and Shiraishi, N., "Rigid polyurethane foams from combined liquefaction mixtures of wood and starch", Mokuzai Gakkaishi, Vol. 41, Pg. 659, 1995
- [52] Li, Y., and Ragauskas, A. J., "Kraft lignin-based rigid polyurethane foam", Journal of Wood Chemistry and Technology, Vol. 32, No. 3, Pg. 210, 2012
- [53] Cateto, C. A., Barreiro, M. F., Ottati, C., Lopretti, M., Rodrigues, A. E., and Belgacem, M. N., "Lignin-based rigid polyurethane foams with improved biodegradation", Vol. 50, No. 1, Pg. 81, 2014
- [54] Bandyopadhyay-Ghosh, S., Ghosh, S. B., and Sain, M., "Synthesis of soy-polyol by two step continuous route and development of soy-based polyurethane foam", Vol. 18, No. 3, Pg. 437, 2010
- [55] Rücker, C., Kümmerer, K., "Modeling and predicting aquatic aerobic biodegradation A review from a user's perspective", Green Chemistry, Vol. 14, No. 4, Pg. 875, 2012
- [56] Matthews, F. L., and Rawlings, R. D., "Composite materials: engineering and science", Alden Press, Oxford, United Kingdom, 1994
- [57] Ivanov, O. N., Valiullin, A. Kh., Ievenko, V. F., and Ioseliani, V. P., "Mechanical properties of a hollow-fiber glass-reinforced plastic", Polymer Mechanics, Vol. 7, No. 2, Pg. 298, 1973
- [58] Céline, Z. M., Nadine, Le B., "Effect of the mineral filler on the surface properties of co-ground polymeric composites", Particle and Particle Systems Characterization, Vol. 24, No. 4-5, Pg. 339, 2007
- [59] Trotignon, J. P., Demdoum, L., and Verdu, J., "Effect of mineral fillers in low concentration on the mechanical properties of polymeric materials. Part 1: Static and fatigue fracture of polypropylene, qualitative aspects", Composites, Vol. 23, No. 5, Pg. 313, 1992

- [60] Chiellini, E., Cinelli, P., Chiellini, F., and Imam, S. H., "Environmentally degradable bio-based polymeric blends and composites", Macromolecular Bioscience, Vol. 4, No. 3, Pg. 218, 2004
- [61] Chaudhary, D. S., Jollands, M. C., and Cser, F., "Recycling rice hull ash: A filler material for polymeric composites?", Advances in Polymer Technology, Vol. 23, No. 2, Pg. 147, 2004
- [62] Bashar, A. S., Khan, M. A., Ali, K. M. I., "Bamboo-plastic composite with MMA", Polymer – Plastics Technology and Engineering, Vol. 35, No. 4, Pg. 581, 1996
- [63] Bisanda, E. T. N., "Sisal-fibre-reinforced composites", PhD thesis, University of Bath, United Kingdom, 1991
- [64] Aji, I. S., Sapuan, S. M., Zainudin, E. S., and Abdan, K., "Kenaf fibres as reinforcement for polymeric composites: A review", International Journal of Mechanical and Materials Engineering, Vol. 4, No. 3, Pg. 239, 2009
- [65] Kokta, B. V., Chen, R. D., Claude, V., and Jacques, L., "Use of wood fibers in thermoplastic composites", Polymer Composites, Vol. 4, No. 4, Pg. 229, 1983
- [66] Netravali, A. N., and Chabba, S., "Composites get greener", Materials Today, Vol. 6, No. 4, Pg. 22, 2003
- [67] Matuana, L. M., Park, C. B., and Bauvhnecz, J. J., "Cell morphology and property relationships of microcellular foamed PVC/wood-fiber composites", Polymer Engineering and Science, Vol. 38, No. 11, Pg. 1862, 1998
- [68] Yoon, J. D., Kuboki, T., Jung, P. U., Wang, J., and Park, C. B., "Injection molding of wood-fiber/plastic composite foams", Composite Interfaces, Vol. 16, No. 7-9, Pg. 797, 2009
- [69] Zhang, H., Rizvi, G. M., and Park, C. B., "Development of an extrusion system for producing fine-celled HDPE/wood-fiber composite foams using CO₂ as a blowing agent", Vol. 23, No. 4, Pg. 263, 2004
- [70] Bledzki, A. K., and Faruk, O., "Microcellular wood fibre reinforced PP composites. International polymer processing", Vol. 21, No. 3, Pg. 256, 2006
- [71] Bledzki, A. K., and Faruk, O., "Effects of the chemical foaming agents, injection parameters, and melt-flow index on the microstructure and mechanical properties of microcellular injection-molded wood-fiber/polypropylene composites", Journal of Applied Polymer Science, Vol. 97, No. 3, Pg. 1090, 2005
- [72] Zhang, Z. X., Zhang, J., Lu, B. X., Xin, Z. X., Kang, C. K., and Kim, J. K., "Effect of flame retardants on mechanical properties, flammability and foamability of PP/woodfiber composites", Composites: Part B, Engineering, Vol. 43, No. 2, Pg. 150, 2012
- [73] Rachtanapun, P., Selke, S. E. M., and Matuana, L. M., "Microcellular foam of polymer blends of HDPE/PP and their composites with wood fiber", Journal of Applied Polymer Science, Vol. 88, No. 12, Pg. 2842, 2003
- [74] Matuana-Malanda, L., Park, C. B., and Balatinecz, J. J., "Characterization of molecular PVC/woo-fiber", Proceeding of Annual Technical Conference, 1995
- [75] Matuana-Malanda, L., Park, C. B., Balatinecz, J. J., "Production of microcellular foamed PVC/wood-fibre composites: Processing and cell morphology relationship", Journal of Engineering and Applied Science, Vol. 2, Pg. 1900, 1996
- [76] Matuana, L. M., Park, C. B., and Balatinecz, J. J., "Effect of cell morphology on the properties of microcellular foamed PVC/wood-fiber composites", American Society of Mechanical Engineers, Material Division (Publication), Vol. 74, Pg. 199, 1996
- [77] Rizvi, G., Matuana, L. M., and Park, C. B., "Foaming of PS/wood fiber composites using moisture as a blowing agent", Polymer Engineering and Science, Vol. 40, No. 10, Pg. 2124, 2000
- [78] Guan, L. T., Yang, C. L., Yuan, N. X., and Xie, D. W., "Foaming of PPC/wood fiber composite using moisture contained in wood as blowing agent", Applied Mechanics and Materials, Vol. 415, Pg. 631, 2013
- [79] Neagu, R.C., Cuénoud, M., Berthold, F., Bounrban, P. E., Gamstedt, E. K., Lindström, M., and Månson, J. A. E., "The potential of wood fibers as reinforcement in cellular biopolymers", Journal of Cellular Plastics, Vol. 48, No. 1, Pg. 71, 2012
- [80] Guo, G., Lee, Y. H., Rizvi, G. M., Park, C. B., "Influence of wood fiber size on extrusion foaming of wood fiber/HDPE composites", Journal of Applied Polymer Science, Vol. 107, No. 6, Pg. 3505, 2008
- [81] Khazabi, M., Gu, R., and Sain, M., "Fiber reinforced soy-based polyurethane spray foam insulation. Part 1: Cell morphologies", BioResources, Vol. 6, No. 4, Pg. 3757, 2011
- [82] Gu, R., Khazabi, M., and Sain, M., "Fiber reinforced soy-based polyurethane spray foam insulation. Part 2: Thermal and mechanical properties", BioResources, Vol. 6, No. 4, Pg. 3775, 2011
- [83] Gu, R., and Sain, M. M., "Effects of wood fiber and microclay on the performance of soy based polyurethane foams", Journal of Polymers and the Environment, Vol. 21, No. 1, Pg. 30, 2013
- [84] Silva, M. C., Takahashi, J. A., Chaussy, D., Belgacem, M. N., and Silva, G. G., "Composites of rigid polyurethane foam and cellulose fiber residue", Journal of Applied Polymer Science, Vol. 117, No. 6, Pg. 3665, 2010
- [85] Yuan, J., and Shi, S. Q., "Effect of the addition of wood flours on the properties of rigid polyurethane foam", Journal of Applied Polymer Science, Vol. 113, No. 5, Pg. 2902, 2009
- [86] Zhu, M., Bandyopadhyay-Ghosh, S., Khazabi, M., Cai, H., Correa, C., and Sain, M., "Reinforcement of soy polyol-based rigid polyurethane foams by cellulose microfibers and nanoclays", Journal of Applied Polymer Science, Vol. 124, No. 6, Pg. 4702, 2012
- [87] Liang, K., and Shi, S. Q., "Nanoclay filled soy-based polyurethane foam", Journal of Applied Polymer Science, Vol. 119, No. 3, Pg. 1857, 2011
- [88] Gu, R., Sain, M. M., and Konar, S. K., "A feasibility study of polyurethane composite foam with added hardwood pulp", Industrial Crops and Products, Vol. 42, No. 1, Pg. 273, 2013
- [89] Chiang, J., Cheng, K., Mendez, S., Ko, Y. F., and Bahr, B., "Developing hybrid biofoams by incorporating hay straw and biodiesel waste glycerol into polyurethane foam", Proceeding of International SAMPE Technical Conference, 2012
- [90] Banik, I., and Sain, M. M., "Role of refined paper fiber on structure of water blown soy polyol based polyurethane foams", Journal of Reinforced Plastics and Composites, Vol. 27, No. 14, Pg. 1515, 2008
- [91] Aranguren, M. I., Rácz, I., and Marcovich, N.E., "Microfoams based on castor oil polyurethanes and vegetable fibers", Journal of Applied Polymer Science, Vol. 105, No. 5, Pg. 2791, 2007

- [92] Han, D. S., Park, I. B., Kim, M. H., Noh, B. J., Kim, W. S., and Lee, J. M., "The effects of glass fiber reinforcement on the mechanical behavior of polyurethane foam", Journal of Mechanical Science and Technology, Vol. 24, No. 1, Pg. 263, 2010
- [93] Yakushin, V., Stirna, U., Bel'Kova, L., Deme, L., and Sevastyanova, I., "Properties of rigid polyurethane foams filled with milled carbon fibers", Mechanics of Composite Materials, Vol. 46, No. 6, Pg. 679, 2011
- [94] Li, Y., "Effect of coupling agent concentration, fiber content, and size on mechanical properties of wood/HDPE composites", International Journal of Polymeric Materials and Polymeric Biomaterials, Vol. 61, No. 11, Pg. 882, 2012
- [95] Bledzki, A. K., and Faruk, O., "Wood fiber reinforced polypropylene composites: Effect of fibre geometry and coupling agent on physico-mechanical properties", Applied Composite Materials, Vol. 10, No. 6, Pg. 365, 2003
- [96] Jiang, H., and Kamdem, D. P., "Development of poly(vinyl chloride)/wood composites. A literature review", Journal of Vinyl and Additive Technology, Vol. 10, No. 2, Pg. 59, 2004
- [97] Maldas, D., Kokta, B. V., and Daneault, C., "Influence of coupling agents and treatments on the mechanical properties of cellulose fiber-polystyrene composites", Vol. 37, No. 3, Pg. 751, 1989
- [98] Sundar, S. T., "Chemical modification of wood fiber to enhance the interface between wood and polymer in wood plastic composites", Master Thesis, University of Idaho, United States, 2005
- [99] Faruk, O, Bledzki, A. K., Fink, H. P., and Sain, M., "Biocomposites reinforced with natural fibers: 2000-2010", Progress in Polymer Science, Vol. 37, No. 11, Pg. 1552, 2012
- [100]Gwon, J. G., Lee, S. Y., Chun, S. J., Doh, G. H., and Kim, J. H., "Effect of chemical treatments of wood fibers on the physical strength of polypropylene based composites", Korean Journal of Chemical Engineering, Vol. 27, No. 2, Pg. 651, 2010
- [101]Kalia, S., Kaith, B. S., and Kaur, I., "Pretreatments of natural fibers and their application as reinforcing material in polymer composites a review", Polymer Engineering and Science, Vol. 49, No. 7, Pg. 1253, 2009
- [102]Lee, Y. H., Sain, M. M., Kuboki, T., and Park, C. B., "Influence of interfacial interation on the foamability of wood fiber/HDPE composites", Proceeding of SAE Technical Conference, 2007
- [103]Bledzki, A. K, and Faruk, O., "Microcellular wood fibre reinforced polypropylene composites in an injection moulding process", Vol. 21, No. 6, Pg. 417, 2002
- [104]Benning, C. J., "Plastic Foams: The Physics and Chemistry of Product Performance and Process Technology", Wiley-interscience, New York, United States, 1969
- [105] Choe, K. H., Soo, D. L., Seo, W. J., and Kim, W. N., "Properties of rigid polyurethane foams with blowing agents and catalysts", Polymer Journal, Vol. 36, No. 5, Pg. 368, 2004
- [106]Seo, W. J., Park, J. H., Sung, Y. T., Hwang, D. H., Kim, W. N., and Lee, H. S., "Properties of water-blown rigid polyurethane foams with reactivity of raw materials", Journal of Applied Polymer Science, Vol. 93, No. 5, Pg. 2334, 2004
- [107] Ema, M., "Reproductive and developmental toxicity of organotin compounds", Bulletin of National Institute of Health Sciences, Vol. 125, Pg. 35, 2007

- [108]Pengjam, W., Saengfak, B., Ekgasit, S., and Chantarasiri, N., "Copper-amine complexes as new catalysts for rigid polyurethane foam preparations", Vol. 123, No. 6, Pg. 3520, 2012
- [109]Kazayawoko, M., Balatinecz, J. J., Woodhams, R. T., and Law, S., "Effects of wood fiber surface chemistry on the mechanical properties of wood fiber-polypropylene composites", International Journal of Polymeric Materials, Vol. 37, No. 2, Pg. 237
- [110]Banik, I., and Sain, M. M., "Water-blown soy polyol based polyurethane foams modified by cellulosic materials obtained from different sources", Journal of Applied Polymer Science, Vol. 112, No. 4, Pg. 1974, 2009
- [111]Singh, H., Sharma, T. P., and Jain, A. K., "Reactivity of the raw materials and their effects on the structure and properties of rigid polyurethane foams", Journal of Applied Polymer Science, Vol. 106, No. 2, Pg. 1014, 2007
- [112] Maris, R. V., Tamano, Y., Yoshimura, H., and Gay, K. M., "Polyurethane catalysis by tertiary amines", Journal of Cellular Plastics, Vol. 41, No. 4, Pg. 305, 2005
- [113]Rashmi, B. J., Rusu, D., Prashantha, K., Lacrampe, M. F., and Krawczak, P., "Development of water-blown bio-based thermoplastic polyurethane foams using bioderived chain extender", Journal of Applied Polymer Science, Vol. 128, No. 1, Pg. 292, 2013
- [114] Wang, T. P., Zheng, Z. M., and Mao, Z. H., "Characterization of polyurethane foams prepared from non-pretreated liquefied corn stover with PAPI", Canadian Journal of Chemical Engineering, Vol. 89, No. 1, Pg. 171, 2011
- [115]Goto, A., Yamaguchi, K., and Hamada, H., "Effect of deformational behavior of cell structure on mechanical properties of polyurethane foam under compressive loading", Proceeding of Annual Technical Conference, 2004
- [116]Zhang, X. D., Macosko, C. W., Davis, H. T., Nikolov, A. D., and Wasan, D. T., "Role of silicone surfactant in flexible polyurethane foam", Journal of Colloid and Interface Science, Vol. 215, No. 2, Pg. 270, 1999
- [117] Campanella, A., Bonnaillie, L. M., Wool, R. P., "Polyurethane foams from soyoilbased polyols", Journal of Applied Polymer Science, Vol. 112, No. 4, Pg. 2567, 2009
- [118] Sipaut, C. S., Ahmad, N., Adnan, R., Rahman, I. A., and Ibrahim, M. N. M., "Effects of starting material and reaction temperature on the morphology and physical properties of polyurethane foams", Cellular Polymers, Vol. 29, No. 1, Pg. 1, 2010
- [120] Seeler, K. A., and Kumar, V., "Tension-tension fatigue of microcellular polycarbonate: initial results", Journal of Reinforced Plastics and Composites, Vol. 12, No. 3, Pg. 359, 1993
- [121]Collias, D. I., Baird, D. G., and Borggreve, R. J. M., "Impact toughening of polycarbonate by microcellular foaming", Polymer, Vol. 35, No. 18, Pg. 3978, 1994
- [122] Weyenberg, I. V., Truong, C., Vangrimde, T., and Verpoest, I., "Improving the properties of UD flax fibre reinforced composites by applying an alkaline fibre treatment", Composites Part A: Applied Science and Manufacturing, Vol. 37, No. 9, Pg. 1368, 2006
- [123]EI-Shekeil, Y. A., Sapuan, S. M., Khalina, A.,Zainudin, E. S., and AI-Shunja'a, O. M., "Influence of chemical treatment on the tensile properties of kenaf fiber reinforced thermoplastic polyurethane coomposite", Express Polymer Letters, Vol. 6, No. 12, Pg. 1032, 2012

[124]Cordeiroa, N., Ornelasa, M., Ashorib, A., Sheshmanic, S., and Norouzic, H., "Investigation on the surface properties of chemically modified natural fibers using inverse gas chromatography", Carbohydrate Polymers, Vol. 87, No. 4, Pg. 2367, 2012